

JGR Earth Surface

FEATURE ARTICLE

10.1029/2018JF004873



Key Points:

- Outlet glacier changes are heterogeneous and result in large uncertainties in future sea-level rise contribution from Greenland
- Uncertain understanding of outlet glacier changes is largely due to ice-ocean and ice sheet basal processes
- Future research needs include expanded observations, improved modeling, and greater inclusion of new researchers

Correspondence to:

G. Catania, gcatania@ig.utexas.edu

Citation:

Catania G.A., L.A Stearns, T. Moon, E. Enderlin, & R.H. Jackson. (2020). Future evolution of greenland's marine-terminating outlet glaciers. Journal of Geophysical Research: Earth Surface, 125, e2018JF004873. https:// doi.org/10.1029/2018JF004873

Received 6 AUG 2019 Accepted 22 NOV 2019 Accepted article online 10 DEC 2019

©2019. American Geophysical Union. All Rights Reserved.

Future Evolution of Greenland's Marine-Terminating Outlet Glaciers

G. A. Catania^{1,2}, L. A. Stearns³, T. A. Moon^{4,5}, E. M. Enderlin⁶, and R. H. Jackson⁷

¹Institute for Geophysics, University of Texas at Austin, Austin, TX, ²Department of Geological Sciences, University of Texas at Austin, Austin, TX, ³Department of Geology, University of Kansas, Lawrence, KS, ⁴National Snow and Ice Data Center, University of Colorado Boulder, Boulder, CO, ⁵Cooperative Institute for Research in Environmental Sciences, University of Colorado Boulder, Boulder, CO, ⁶Department of Geosciences, Boise State University, Boise, ID, ⁷Department of Marine and Coastal Sciences, Rutgers University, New Brunswick, NJ

Abstract Mass loss from the Greenland ice sheet (GrIS) has increased over the last two decades in response to changes in global climate, motivating the scientific community to question how the GrIS will contribute to sea-level rise on timescales that are relevant to coastal communities. Observations also indicate that the impact of a melting GrIS extends beyond sea-level rise, including changes to ocean properties and circulation, nutrient and sediment cycling, and ecosystem function. Unfortunately, despite the rapid growth of interest in GrIS mass loss and its impacts, we still lack the ability to confidently predict the rate of future mass loss and the full impacts of this mass loss on the globe. Uncertainty in GrIS mass loss projections in part stems from the nonlinear response of the ice sheet to climate forcing, with many processes at play that influence how mass is lost. This is particularly true for outlet glaciers in Greenland that terminate in the ocean because their flow is strongly controlled by multiple processes that alter their boundary conditions at the ice-atmosphere, ice-ocean, and ice-bed interfaces. Many of these processes change on a range of overlapping timescales and are challenging to observe, making them difficult to understand and thus missing in prognostic ice sheet/climate models. For example, recent (beginning in the late 1990s) mass loss via outlet glaciers has been attributed primarily to changing ice-ocean interactions, driven by both oceanic and atmospheric warming, but the exact mechanisms controlling the onset of glacier retreat and the processes that regulate the amount of retreat remain uncertain. Here we review the progress in understanding GrIS outlet glacier sensitivity to climate change, how mass loss has changed over time, and how our understanding has evolved as observational capacity expanded. Although many processes are far better understood than they were even a decade ago, fundamental gaps in our understanding of certain processes remain. These gaps impede our ability to understand past changes in dynamics and to make more accurate mass loss projections under future climate change. As such, there is a pressing need for (1) improved, long-term observations at the ice-ocean and ice-bed boundaries, (2) more observationally constrained numerical ice flow models that are coupled to atmosphere and ocean models, and (3) continued development of a collaborative and interdisciplinary scientific community.

Plain Language Summary Increasing mass loss from the Greenland ice sheet (GrIS) in response to changes in global climate has motivated the scientific community to understand how much sea level rise will happen in the coming decades. Observations now indicate that the impact of a melting GrIS are more widespread than just sea-level rise and include changes to ocean properties and circulation, nutrient and sediment cycling, and ecosystem function. Major uncertainties still hamper accurate predictions of these impacts, particularly for outlet glaciers in Greenland that terminate in the ocean because their flow is strongly controlled by multiple processes that alter their boundary conditions at the ice-atmosphere, ice-ocean, and ice-bed interfaces. Many of these processes change on a range of overlapping timescales and are challenging to observe. Here we review the scientific progress in understanding how GrIS outlet glaciers respond to climate and how our understanding has changed over time as observations have increased. We conclude with recommendations for (1) improved, long-term observations at the ice-ocean and ice-bed boundaries, (2) more observationally-constrained ice flow models that are linked to atmosphere and ocean models, and (3) continued development of a collaborative and interdisciplinary scientific community.



1. The Importance of Greenland's Outlet Glaciers to Society

The Greenland ice sheet (GrIS) is the second largest reservoir of land-based ice in the world with $\sim 2 \times 10^6$ km³ of ice and the ability to increase global sea level by 7.42 m (Morlighem et al., 2017). The total ice sheet mass balance is an important measure of its "health" with two interconnected components contributing to mass balance changes: (1) solid ice discharge (through iceberg calving) via dynamic changes that occur as a result of atmospheric, oceanic, and other processes and (2) ice surface mass balance (SMB) changes in response to changing atmospheric conditions (see Glossary). While surface accumulation within the interior areas of the GrIS has increased slightly in recent years (Noël et al., 2018), large mass loss is concentrated around the periphery of the GrIS. This mass loss includes both large surface mass losses and changes to marine-terminating outlet glaciers, which act as rapid mass conveyor belts moving ice from the interior to the ocean.

Recent observations suggest that during 1972–2018, ~ 66% of the GrIS mass loss was due to dynamic change associated with marine-terminating outlet glaciers, producing a ~9-mm increase in sea level, with pronounced dynamic losses in NW and SE Greenland and roughly half of this sea level increase occurring in the last 8 years (Mouginot et al., 2019). While estimates of past mass contribution to sea level have low uncertainties, multiple difficulties remain in projecting future GrIS mass change. In part, there is large uncertainty because future human behavior and climate policy is unknown. There also remains a debate about the physics that dominate important processes contributing to dynamic changes, particularly for outlet glaciers that connect to the ocean (Minchew et al., 2019; Pattyn, 2018; Stearns & van der Veen, 2018; Stearns & van der Veen, 2019; Sutherland et al., 2019). In addition, while most of the outlet glacier dynamic response occurs in the terminal zone (at the ocean boundary), contemporary terminus change is proven long-lasting, ultimately affecting the full ice sheet, and is responsible for much of the expected sea-level rise from Green-land (Felikson et al., 2017; Howat et al., 2008; Price et al., 2011). Thus, it is important to understand more than what happens at the ice sheet marine margin alone.

Here we present an overview of the Greenland marine-terminating outlet glacier settings and controls, examined through the lens of observational and modeling capacity growth in recent decades and a view of future needs. This review builds on existing reviews of glacier calving and crevasse mechanisms (Benn et al., 2007; Colgan et al., 2016), submarine melt (Benn et al., 2017; Straneo & Cenedese, 2015; Truffer & Motyka, 2016), and numerous studies that examine the Greenland climate forcing (Fyke et al., 2018; Straneo & Heimbach, 2013) and the heterogeneous outlet glacier response (Carr et al., 2013). The study of Greenland outlet glaciers is especially urgent given that the recent rapid climate warming underway in the Arctic (see section 3) and concurrent rapid changes in GrIS mass loss (Mouginot et al., 2019) portend a strong response in GrIS mass change.

Rapid expansion of observational capabilities, both in-situ and via satellites, has improved our understanding of outlet glacier changes (see section 4) and the processes responsible (see section 5). Recent attention has focused on ocean forcing, but the response of outlet glaciers to climate forcing is not straightforward. A complex picture has emerged, with interwoven processes controlling outlet glacier dynamics and with outlet glaciers playing a multidimensional role in the interconnected Arctic ice-ocean-climate-biological system (section 6). Given that Greenland's outlet glaciers are important at all scales—locally (e.g., biological productivity and nutrient cycling), regionally (e.g., fjord circulation and ocean stratification), and globally (e.g., ocean thermohaline circulation and sea-level rise)—we conclude with recommendations for future research (see section 7) to expand our understanding of outlet glacier change and associated effects.

2. Context and Setting

2.1. Greenlandic Marine-Terminating Outlet Glacier Characteristics

Outlet glaciers act as arteries for ice flux from the ice sheet interior; their topographically controlled high rates of ice flow allow them to drain interior ice to the marine margin where they terminate in the ocean. Across the globe, the character of marine-terminating outlet glaciers spans a range from the wide and deep, ice-shelf-buttressed outlet glaciers of Antarctica to the thin, well-grounded and relatively warm glaciers of Svalbard, Alaska, and other Arctic regions. Greenland's glaciers fill a unique central range on this spectrum, with many narrow, deeply grounded outlet glaciers and examples both of ice-shelf-buttressed outlets (in the north) and thin, well-grounded glaciers (e.g., much of the east). Most GrIS outlet glaciers have bedrock mar-





Figure 1. The Greenland Ice Sheet situated in the Arctic showing ocean currents. The size of ocean current arrows indicates water mass; color of arrows indicates heat transport. Surface ice flow speed from (Joughin et al., 2010) highlights the numerous fast-moving outlet glaciers around the periphery of the ice sheet that drain ice from the interior. Purple triangles indicate outlet glaciers identified with flow rates above 50 m/year. Bathymetry data from the GEBCO Grid (GEBCO Compilation Group, 2019) show the deep troughs created on the sea floor from the extent of past ice streams to the continental slope.

gins and deep central troughs carved through erosion, similar to their marine-terminating mountain-glacier counterparts in Alaska and the Antarctic Peninsula. However, some outlet glaciers in Greenland have fast surface flow speeds that reach far into the ice sheet interior (e.g., Jakobshavn Isbrae and the Northeast Greenland Ice Stream) with similarities to the large ice streams draining the Antarctic ice sheet. For these glaciers, lateral boundaries of fast flow are constrained by slow-moving ice in the interior and bedrock towards the ice sheet margin. Greenland's outlet glaciers are narrower than their Antarctic counterparts, but compared to outlet glaciers in the rest of the Arctic, many GrIS outlet glaciers are deeper and connected to a much larger supply of inland ice.

While increasing air and ocean temperatures are driving higher mass loss from outlet glaciers around the globe, the unique character of climate forcing in different regions alters outlet glacier dynamic response. In the Antarctic Peninsula, solid ice discharge via calving is the dominant outlet glacier ice loss process, likely exacerbated by ice shelf weakening from surface and submarine melt (Glasser & Scambos, 2008; Khazendar et al., 2007; Scambos et al., 2004, 2009). Other regions around Antarctica (like the Amundsen Embayment) show large concentrations of dynamic mass loss associated with warm subsurface ocean temperatures suggesting that, like Greenland, the ocean exerts a strong influence on Antarctic mass loss (Cook et al., 2016; The IMBIE team, 2018). Outside of the GrIS, widespread retreat and dynamic acceleration is evident across the Arctic, though SMB is a more dominant contributor to mass loss than dynamics. Changes in discharge

only account for $\sim 20\%$ of mass loss outside the GrIS (McNabb et al., 2015; Van Wychen et al., 2016). While the ocean likely plays an important role for some glaciers outside the GrIS, most of these systems are more responsive to atmospheric changes because they are well-grounded or have shallow regions of ocean contact (Cook et al., 2019).

The GrIS contains ~280 fast-flowing (>100 m/year) marine-terminating glaciers (Figure 1) with a high degree of heterogeneity across a range of parameters (Mankoff et al., 2019). Ice discharge from these glaciers is ~500 Gt/year total, with large variations in individual glacier ice discharge associated with interglacier differences in width (1–30 km), thickness (~100–2,000 m), terminus basal conditions (grounded, partially floating, fully floating), terminus conditions (open water, mélange—a granular matrix of icebergs, bergy bits, and sea ice—or ice shelf presence), basal substrate (bedrock, sediments, water), and other topographic controls (Enderlin et al., 2014; King et al., 2018; Mankoff et al., 2019). In the far north and northeast, there are relatively few outlet glaciers; most of which terminate in perennial floating ice shelves. In the northwest, ice flow is less restricted by topography further inland, and there exists a larger abundance of narrower outlet glaciers draining the ice sheet. Here many glaciers develop ephemeral floating termini each winter, which are lost when calving rates increase each spring. Southeastern and western outlet glaciers typically lack floating ice shelves and are fully grounded at their termini, with the exception of the largest glaciers (Enderlin & Howat, 2013; James et al., 2014).

3. Climate Setting in Greenland

The last several decades have seen substantial changes in atmospheric and oceanic conditions around Greenland (Bevis et al., 2019; Straneo & Heimbach, 2013). Large-scale and multidecadal processes as well as local and short-term changes in natural and anthropogenic forcings influence outlet glaciers. Recent rapid Greenland mass loss cannot, however, be explained by natural variability alone. Current changes are closely aligned with significant acceleration in anthropogenic warming, which is altering small to large-scale processes critical to ice sheet mass balance (Aschwanden et al., 2019).

Large-scale atmospheric variations influence surface mass balance across the full ice sheet. Annual to multidecadal GrIS SMB correlates, at varying significance, with the North Atlantic oscillation (NAO), Greenland blocking index (GBI), and Atlantic multidecadal oscillation (AMO) (Hanna et al., 2012, 2016). A negative summer NAO phase is associated with high annual mass losses via enhanced summertime warming and reduced snowfall, particularly in western Greenland (Bevis et al., 2019). High values of the GBI, which measure and describe mean 500 hPa geopotential height for the 60–80° N, 20–80° W region, are also associated with large, ice-sheet-wide melt events (Delhasse et al., 2018; Hanna et al., 2013; Hofer et al., 2017). The decadal-scale periodicity of the NAO and GBI, and the approximately 60-year periodicity of the AMO, are evident in multicentury Greenland ice cores (Trusel et al., 2018). The summer NAO and annual surface mass balance have been predominantly negative since the early 2000s (van den Broeke & Lenaerts, 2014), punctuated by a strong positive index and a pause in surface mass loss in 2013 (Bevis et al., 2019). Although the summer GBI was generally slightly negative from the 1960s–1980s, it increased throughout the 1990s–2000s, reaching a peak during the extreme melt season in 2012 (Hanna et al., 2016). Since the late 1990s, the AMO has been in a positive phase that has been amplified by global warming and currently exceeds AMO values during the period of Greenland warming in the 1930s and 1940s (Enfield et al., 2001).

Along with approximately decadal-scale atmospheric forcing, shorter-term variations are also important. For example, commonly occurring low-level clouds can enhance downwelling infrared flux without obstructing solar radiation, inducing short-term melt events (Bennartz et al., 2013). Atmospheric rivers, which can transport warm, moist air from North America can also enhance short-term melt events, particularly in western Greenland (Mattingly et al., 2018). Short-term variations may also be affected by the broad-scale atmospheric state. For example, the occurrence of atmospheric rivers over Greenland has increased since the late 1990s, in part connected to the negative NAO phase (Mattingly et al., 2018).

On the oceanic side, GrIS outlet glaciers are influenced by ocean properties (e.g., temperature, salinity, and velocity) at the ice-ocean interface, and these ocean properties are in turn modulated by regional ocean circulation around Greenland. (Details of the oceanographic controls on Greenland's outlet glaciers are covered in the review papers of Straneo and Heimbach (2013) and Straneo and Cenedese (2015).) Greenland lies at the confluence of the export pathway of waters from the Arctic and the recirculation of subtropical Atlantic Waters (AW). The East Greenland Coastal Current carries cold, fresh polar-origin water (PW) southward



along the east coast and meets the Irminger Current south of Denmark Strait (Figure 1). As an extension of the North Atlantic Current, the Irminger Current forms a key subpolar gyre boundary current and carries warm AW toward Greenland. Together, the merged East Greenland-Irminger Current travels along the southeast coast, wraps around the southern end of Greenland at Cape Farewell, and continues northward as the West Greenland Current towards Baffin Bay.

Transported via this system of currents, AW is an important source of heat for outlet glaciers. On the Greenland continental shelves, the colder PW is generally found in the surface layer (upper ~200 m) and near the coast, blocking AW surface pathways towards glaciers. However, AW is typically found below and offshore of PW on the shelf (Sutherland & Pickart, 2008) and fills the deep basins within most Greenlandic fjords (Straneo et al., 2011, 2012). The relative offshore thickness of PW and AW layers can modulate water mass access to outlet glaciers, influencing ice-ocean interface melting. For example, in the mid-1990s, the AW layer thickened offshore of Greenland (Våge et al., 2011), possibly in response to combined upper ocean warming since the 1950s (Levitus & Antonov, 2005) and decadal-scale natural ocean variability (Straneo & Heimbach, 2013). This AW layer thickening might have increased the ocean heat available at glacier termini and enhanced submarine melting, as suggested by a correlation between large-scale ocean properties and glacier behavior (Holland et al., 2008; Straneo & Heimbach, 2013). However, there are many critical but poorly resolved links between large-scale ocean variability and glacier response, including fjord circulation and submarine melt dynamics.

To influence submarine melting at glacier termini, Atlantic-origin waters must cross the continental shelf, transit through fjords, and come into contact with glacier termini. Bathymetry is one important control on ocean heat transport towards glacier termini. On the continental shelf, deep valleys or troughs—carved by the past expansion of glaciers during the Last Glacial Maximum (O'Cofaigh et al., 2013)— form a pathway for warm AW to cross the shelf and reach the fjord mouth. Fjords, in turn, form deep, narrow conduits between the ocean at the shelf and the glacier termini. Some fjords contain bathymetric sills (most commonly at the mouth) that can limit the encroachment of warm waters at depth, likely making glaciers within those fjords less vulnerable to ocean warming (Carroll et al., 2016; Millan et al., 2018).

Other fjord circulation processes—including freshwater buoyancy forcing, atmospheric forcing, tides, and coastal-trapped waves—also control ocean heat transport towards the glacier and glacial freshwater export out to the coastal ocean. During the summer, subglacial meltwater discharges at the outlet glacier grounding line, driving vigorous upwelling plumes and setting up a fjord exchange flow that helps draws AW toward glacier termini (Carroll et al., 2015; Motyka et al., 2003; Rignot et al., 2010; Sciascia et al., 2013). Other freshwater sources, including surface runoff and submarine glacier and iceberg melting, also contribute to the buoyancy forcing that drives a fjord exchange flow. In addition, wind forcing—both locally within the fjord and remotely on the shelf—can drive energetic fjord flows and vigorous exchange between fjords and shelf ocean (Carroll et al., 2017; Cowton et al., 2016; Fraser & Inall, 2018; Jackson et al., 2014, 2018; Spall et al., 2017). These wind events, often associated with atmospheric low pressure systems, are strongest during winter and have a seasonal cycle that is opposite to the seasonality of freshwater forcing (Harden et al., 2011; Oltmanns et al., 2014). Freshwater forcing, wind forcing, and other fjord processes (tides, internal waves, etc.) all contribute to the net exchange of heat, salt, and freshwater between the shelf and the fjord and in combination control the near-glacier ocean properties that influence the GrIS.

4. Evolving Understanding of GrIS Outlet Glaciers

Despite nearly a century of ice sheet observations (Bjørk et al., 2012; Gabel-Jorgensen, 1935), only in recent decades has the importance of GrIS outlet glaciers been fully recognized. This is largely due to the improved spatial and temporal resolution of observations from air and space-borne instruments, particularly since the 1990s (Figure 2). These observational improvements have been proven pivotal for quantifying the mass balance state of the ice sheet and for improving our understanding of outlet glacier controls. As an example, the ice sheet was thought to be close to a balanced state during the 1970s and 1980s but, until recently, we lacked the observational data to demonstrate this within measurement uncertainties (Mouginot et al., 2019).

In the 1990s, NASA's Program for Arctic Regional Climate Assessment was the seed for several observational ground and air campaigns aimed at improving ice sheet mass balance estimates. NASA's Airborne Topographic Mapper began flying over Greenland with regularity during this time and revealed peripheral thinning rates in excess of 1 m/year (Frederick et al., 2000; Krabill et al., 2004), which were subsequently



Figure 2. Qualitative interpretation of progress in the field of outlet glaciers in Greenland showing rapid gains in observations (both sensor type and scales have expanded), understanding of important mechanisms, and model capabilities with time.

validated by NASA's ICESat mission launched in 2003 (Pritchard et al., 2009). Although the relative contributions of the observed thinning from ice dynamics or surface mass balance changes were unknown at that time, more recent reconstructions of terminus position (Catania et al., 2018; Howat & Eddy, 2011; Murray et al., 2015) and discharge (Mouginot et al., 2019; Rignot & Kanagaratnam, 2006) support the hypothesis that the onset of dynamic flow acceleration occurred in the 1990s. These changes coincided with increases in atmospheric warmth and oceanic heat content as described above.

Subsequently, direct monthly GrIS mass estimates became available from NASA's Gravity Recovery and Climate Experiment (GRACE) satellite launched in 2002, producing, for the first time, an ice-sheet-wide picture of the spatio-temporal mass balance pattern. As GRACE data accumulated, complex and time-varying mass loss patterns were revealed. Mass loss was concentrated in the southeast prior to ~2007, but notable acceleration of mass loss into the northwest ultimately allowed the northwest to contribute ~70% of GrIS mass loss during 2003-2013 (Chen et al., 2011; Khan et al., 2010; Velicogna et al., 2014). GRACE data also enabled scientists to fully appreciate the rapidity of GrIS changes. Even seasonal and shorter duration events had an impact on ice mass balance (Velicogna, 2009), meaning that short duration observations could impart bias in mass balance estimates. By comparing regional climate model estimates of ice sheet SMB to the total mass loss data from GRACE, scientists were able to distinguish between SMB and dynamic mass change for the ice sheet on a range of time and space scales (van den Broeke et al., 2016; Velicogna et al., 2014). This work thrust Greenland's outlet glaciers into the spotlight (in terms of both the research priority and the news media) as significant potential contributors to sea level rise.

With a focus on outlet glaciers as an important control of GrIS mass balance, researchers began to identify and understand the underlying mechanisms for the observed changes. While changes to dynamic regime (e.g., stable to retreating) were likely initiated from the ocean (Howat & Eddy, 2011; Murray et al., 2010; Nick et al., 2009; Seale et al., 2011; Straneo et al., 2010), the exact mechanism for ocean-induced retreat remained elusive for several years. For many outlet glaciers, speedup and thinning were hallmark indicators



of dynamic change, demonstrating the sensitivity of outlet glacier dynamics to changes in terminus perturbations (Howat et al., 2008; Joughin et al., 2008). As observations expanded in time and space, however, this simple model of glacier dynamic response to climate perturbation became more complicated. Not all glaciers responded to climate perturbations in the same way, and even glaciers in adjacent fjords exhibited vastly different dynamic changes in elevation (Csatho et al., 2014; Felikson et al., 2017; Howat et al., 2007; McFadden et al., 2011), ice speed (Bevan et al., 2012; Howat et al., 2007; Joughin et al., 2010; Moon et al., 2012), terminus position (Bevan et al., 2012; Catania et al., 2018; Howat et al., 2007; McFadden et al., 2011; Moon & Joughin, 2008; Murray, Scharrer, et al., 2015; Warren & Glasser, 1992), and ice discharge (Enderlin et al., 2014; King et al., 2018). This work made obvious the idea that outlet glacier controls may be regionally or locally variable.

During this time, there was also considerable research into the role of ice sheet surface meltwater in altering ice dynamics. Much research focused on land-terminating glaciers (Bartholomew et al., 2011; Das et al., 2008; Joughin et al., 2008b; Sole et al., 2013; Zwally et al., 2002) because of the relative ease of making observations in these regions compared to heavily crevassed outlet glaciers. This work revealed the ice velocity response to subglacial surface meltwater input on a range of time scales. While ice velocity increases rapidly in response to changes in subglacial water storage (Bartholomaus et al., 2008), there are self-regulation processes that permit efficiency to be gained in the subglacial system causing ice velocity slow down to occur over longer time scales (Nienow et al., 2017; Tedstone et al., 2015). Similar controls are expected for outlet glaciers, but observations using correlations between surface melt and ice flow speed in outlet glaciers are sparse (De Juan et al., 2010; Moon et al., 2014). As our attention in recent years shifted to the ice-ocean boundary, the important role of subglacially-routed surface meltwater as a driver of submarine melt and terminus retreat, has emerged (Chauché et al., 2014; Fried et al., 2015; Motyka et al., 2003; Rignot et al., 2010; Slater et al., 2015; Straneo et al., 2013) (see section 5.2.2). This somewhat unexpected mechanism by which surface melt influences glacier dynamics illustrates the importance of examining mass loss processes holistically to account for linkages between processes at the subglacial substrate, the ocean and atmospheric boundaries, and the deep interior dynamics of the ice sheet.

This brief history highlights the importance of improved observations—higher resolution (spatial and temporal) and a wider variety of observation type—to our deeper understanding of Greenland's outlet glacier controls. Simultaneous development has also occurred in ice sheet model methodologies, which permit greater data assimilation, higher resolution, improved model physics, and the development of competing models. In combination, these advances support improved projections of future outlet glacier change.

5. Controls on Outlet Glacier Dynamics

While outlet glaciers extend far into the ice sheet interior, much work has focused on the terminus boundary at the ocean because this is where the changes in mass balance have been the largest. GrIS outlet glacier termini exist in an environment where they are influenced by external climatic factors like ocean and atmospheric properties, as well as internal factors like substrate composition, geometric properties, and inland dynamics (Figure 3; Carr, Stokes, and Vieli (2013)). Because changes in these factors can be superimposed on, and interact with, one another, it is often difficult to attribute an observed glacier dynamic change to a particular forcing factor. In general, however, change in terminus position over time arises from an imbalance between ice flux from the interior and frontal ablation (i.e., the sum of iceberg calving (section 5.2.1) and submarine melting (section 5.2.2)). Glaciers are expected to advance into their fjords if they are supplied more ice from the interior than they lose. This could result from ice flux acceleration towards the marine margin, surface snow accumulation in excess of surface melt (section 5.3), or reduced terminus ablation rates. Glaciers are expected to retreat inland when ice flux from the interior does not match the loss of ice via frontal ablation. This could occur from ice flux deceleration, increased calving and/or submarine melting rates, or greater surface melting than snow accumulation. Terminus retreat usually induces outlet glacier thinning, predominantly in downstream regions. This ice sheet surface steepening subsequently induces an ice flow acceleration, but can also induce ice flow convergence, which may counteract acceleration. While this pattern of outlet glacier dynamic change in response to retreat has been observed around the GrIS, significant spatio-temporal heterogeneity in this response complicates our understanding of outlet glacier controls.





Figure 3. Conceptual diagram of outlet glacier processes at play in the terminal zone where the atmosphere, ice sheet, substrate and ocean interact. Red arrow indicates the ocean heat forcing that is at depth in fjords. Black arrow indicates that dynamic changes of outlet glaciers are more pronounced toward the glacier terminus. Blue arrows indicate subglacial water, which upwells at the submarine terminus and rises along the terminus face carrying sediment and nutrients (brown arrows) with it as well. Subglacial plumes, iceberg melt and far-field ocean circulation all act to influence fjord circulation.

5.1. Morphological Controls

The geometry of an outlet glacier (width, surface, and bed topography) exerts a first-order control on dynamics by influencing the force balance that governs ice flow (van der Veen & Whillans, 1989). Here we use the term bed topography to mean the topography beneath the glacier and bathymetry to mean the topography beneath the ocean within an outlet glacier fjord. Morphological controls on GrIS outlet glaciers include geometrical effects and effects related to bed characteristics. This includes the presence or absence of till, till rheology during deformation, and water pressure variations at the ice-bed interface. Little direct observation of bed characteristics exists for the GrIS as a whole, with even fewer in outlet glacier regions owing to the difficulty, and expense, in obtaining such observations. The basal regime beneath outlet glaciers is expected to be thawed (MacGregor et al., 2016), although the exact relationship between bed conditions and ice flow remains elusive (Minchew et al., 2019; Stearns & van der Veen, 2018; Stearns & van der Veen, 2019). Borehole observations for a land-terminating region near the ice sheet edge confirm the presence of liquid water at the base of the ice sheet at high-pressure conditions, enabling fast ice motion by reducing effective pressure at the bed (Andrews et al., 2014). We expect similar processes to occur beneath outlet glaciers. Evidence from seismic experiments also suggests that thick till is present in places (Walter et al., 2014) and is highly porous (up to 40%) and thus weak (Dow et al., 2013; Walter et al., 2014), although this till is not ubiquitous across the ice sheet (Harper et al., 2017).

Initial surveys of basal topography beneath GrIS outlet glaciers (e.g., Jakobshavn Isbrae) revealed that, unlike the ice streams in Antarctica, most GrIS outlet glaciers flow in narrow (<10 km), deeply incised (<3 km) submarine troughs that are laterally constrained by bedrock, with flow driven by steep surface slopes (Clarke & Echelmeyer, 1996; Truffer & Echelmeyer, 2003). Thanks to coordinated, long-term effort to complete extensive ice-penetrating radar surveys, the basal topography has been constrained across a large portion of the GrIS (Bamber et al., 2013; Gogineni et al., 2014; Studinger et al., 2010). Similarly, multibeam surveys within several fjords near glacier termini have been collected to determine fjord bathymetry (Fenty et al., 2016). These and other observations have been combined via the principle of mass conservation to produce a spatially complete GrIS subglacial topography map, including estimates across the terminal zone of many



outlet glaciers (Morlighem et al., 2017). This valuable map has been used to refine our early understanding of the important role of topography on outlet glacier behavior (Hughes, 1975; Mercer, 1978; Weertman, 1961). For example, observations of recent glacier terminus position changes suggest near-terminus topography is a critical determinant of terminus stability (Brough et al., 2019; Bunce et al., 2018; Catania et al., 2018). Although not the case for all glaciers, outlet glacier termini that rest on retrograde slopes of terminal moraines (termed sills in the oceanographic literature) are generally more susceptible to retreat given an external forcing, and, once retreat begins, it may continue unabated until the terminus reaches a region of prograde bed slope (Catania et al., 2018, Schoof et al., 2017; Haseloff and Sergienko, 2018). Uncertainties in bed topography remain in under-sampled regions (Morlighem et al., 2017), however, and because outlet glaciers are sensitive to even small-scale (less than a kilometer scale) topographic variations (Catania et al., 2018; Enderlin et al., 2013), large uncertainties remain in projections of future outlet glacier behavior.

The deep subglacial trough that defines most GrIS outlet glaciers is likely formed through bed erosion over multiple glacial cycles. Deeper troughs are formed in steeper regions of Greenland as a result of greater flow convergence contributing to enhanced erosion (Kessler et al., 2008). This erosion both lowers the bed of outlet glaciers and produces sediment that is transported to the grounding line where it can be deposited in the form of a terminal moraine (Dowdeswell & Vásquez, 2013; Jaeger & Koppes, 2016). The presence of a terminal moraine or sill helps to stabilize the terminus by (1) providing additional flow resistance (Brinkerhoff et al., 2017; Morlighem et al., 2016), (2) limiting the warm water access to the terminus at depth (Bartholomaus et al., 2013), and (3) reducing buoyancy effects on calving (Enderlin et al., 2018; Post et al., 2011). In fact, the presence of a moraine at the glacier terminus exerts such a profound control on dynamics that the coupling between ice and sediment dynamics is considered to be solely responsible for the tidewater glacier cycle, a pattern of slow advance and rapid retreat for marine-terminating glaciers that occurs in the absence of climate forcing (Brinkerhoff et al., 2017; Post et al., 2011). Rapid retreat occurs as a result of instability of the terminus position resting on a retrograde bedslope, as described above. Advance is controlled by, amongst other things, the rate of sediment transport and is thus related to surface melt. The advance process is slow because sufficient mass must be accumulated and transported to the terminus. To advance into an overdeepened fjord area, a glacier must also bulldoze enough sediment into the overdeepening to sufficiently inhibit frontal ablation and promote terminus advance (Brinkerhoff et al., 2017; Motyka et al., 2006; Nick et al., 2009).

Although frictional resistance to flow provided by a terminal moraine/sill can help to stabilize glacier termini, generally little frictional resistance is provided at the glacier bed where the ice is close to flotation (Enderlin, Carrigan, et al., 2018; Shapero et al., 2016; Stearns & van der Veen, 2019). As a result, the variable width of the glacier fjord and the changes in width that accompany terminus movement also impact glacier dynamics (Carr et al., 2013). Enderlin et al. (2013) found that, all else equal, glaciers with wider termini and/or with geometries that widen inland of the terminus are more sensitive to climate perturbations than their narrower counterparts. In fact, inland widening of the fjord may exert stronger control over retreat than the presence or absence of a terminal moraine (Åkesson et al., 2018).

The geometry up-glacier of the terminus region also exerts control over outlet glacier dynamics. Following terminus retreat, interior losses of ice occur as a result of terminus-induced diffusive thinning that propagates into the interior (Price et al., 2011). Diffusive thinning is understood with kinematic wave theory (Nye, 1960; Pfeffer, 2007; van der Veen, 2001), which relates how a glacier dynamically adjusts to perturbations. The extent of inland thinning for individual glaciers was recently shown to correlate to glacier geometry via the Peclet number, a nondimensional number relating the rate of advection of a perturbation at the terminus to the rate of diffusion (Felikson et al., 2017). Thinning limits occur at the heads of outlet glacier submarine troughs where steep bed slopes restrict diffusive thinning from propagating farther inland, placing important limits on the ability of some glaciers to contribute significantly to near-term sea level rise (Felikson et al., 2017). Inland thinning observations following the near-synchronous 2005 retreat of SE Greenland glaciers support a strong diffuse thinning dependence on surface slope: Up to 10s of meters of thinning were observed along the steep, confined, fast-flowing outlet glaciers within years of terminus retreat with much lower rates of thinning across the flatter interior Howat, Joughin, et al. (2008). Estimates of dynamic mass loss from ice sheet models suggest that mass loss due to such diffusive thinning far outweighs mass loss that directly results from terminus retreat, with 75% of estimated sea-level rise as the result of interior thinning (Price et al., 2011).



5.2. Oceanic Controls of Outlet Glaciers 5.2.1. Calving of Icebergs

Calving is a two-step process involving both the fracture and mechanical detachment of icebergs from the glacier (Bassis & Jacobs, 2013). Although the precise mechanisms driving calving may vary in space and time, observations and models of Greenland's glaciers suggest that three major processes influence the calving volume flux: (1) buoyancy at the terminus, (2) buttressing provided by ice mélange, and (3) undercutting via submarine melt (section 5.2.2). The relative importance of these processes likely varies widely with glacier geometry, which broadly controls the ice flux towards the terminus, the stress balance at the terminus, and the access of warm water to the terminus face.

Glaciers occupying deeply incised channels tend to approach buoyant conditions at their termini. Increases in basal water pressure with subglacial channel depth (such that ocean water does not penetrate the glacial hydrologic system) and the hydrostatic pressure imbalance at the terminus support a generally extensional flow regime that promotes buoyancy at the terminus (Murray et al., 2015). Buoyant flexure at the grounding line (Logan et al., 2013) likely supports basal crevasse formation (James et al., 2014; van der Veen, 1998). Basal crevasses can propagate upwards via tidal pumping to connect to existing surface crevasses and create large volume, full-thickness, laterally extensive iceberg calving events (Amundson et al., 2010; Bassis & Jacobs, 2013; Fried et al., 2018; James et al., 2014; Murray, Selmes, et al., 2015). This buoyant-flexure-driven style of calving can disrupt fjord stratification (Burton et al., 2012) and produce icequakes that are large enough to be recorded globally (Ekstrom et al., 2006; Murray, Selmes, et al., 2015) as the iceberg rotates backwards during detachment and scrapes the new terminus face and/or the subglacial substrate (Amundson et al., 2010). Since buoyancy-driven calving typically requires capsize of the full-thickness icebergs that detach from the glacier; the rate of calving via this mechanism is sensitive to buttressing provided by the mélange that occupies many Greenland fjords seasonally to perennially (Amundson et al., 2010; Amundson & Burton, 2018; Fried et al., 2018; Robel, 2017). Glaciers that terminate in deep fjords containing seasonal mélange are prone to larger seasonal oscillations in terminus position that are more directly correlated with changes in mélange presence or strength (Cassotto et al., 2015; Fried et al., 2018; Howat et al., 2011; Moon et al., 2015).

Outlet glaciers that are thinner and well-grounded appear to be less sensitive to mélange presence and more sensitive to terminus melt (section 5.2.2), which can dramatically undercut portions of the terminus face (Fried et al., 2015). Eventually, undercutting becomes sufficiently large and torques the overhanging ice (which is typically heavily crevassed) to drive mechanical failure via iceberg sloughing (Bartholomaus et al., 2012; Fried et al., 2015; O'Leary & Christoffersen, 2013). This calving style produces local terminus embayments associated with regions where subglacial discharge emerges at the grounding line (Chauché et al., 2014; Fried et al., 2018). The amount of seasonal terminus change associated with this calving style appears to be directly controlled by the glacier surface melt volume flux via subglacial discharge (Fried et al., 2018). A number of Greenland's outlet glaciers are thin and calve via sloughing, making this an important calving mechanism.

5.2.2. Submarine Melting of Glacier Termini

Submarine melting of glacier termini—the direct mechanism by which the ocean can influence outlet glaciers—has garnered increased attention as growing evidence points towards a correlation between Greenland glacier dynamics and ocean conditions. For example, benthic foraminiferal data from Disko Bay in western Greenland show a close correlation between Jakobshavn Isbrae terminus positions over the last 100 years and subsurface ocean temperatures (Lloyd et al., 2011). Further, modern observations indicate widespread terminus retreat and thinning occurring when climate conditions promote elevated submarine melt rates (Catania et al., 2018; Holland et al., 2008; Rignot et al., 2016; Straneo & Heimbach, 2013) and slowing and thickening occurring when subsurface ocean waters cool (Khazendar et al., 2019). However, no simple relationship exists between ocean and ice that is shared across glacier-fjord systems in Greenland (Slater et al., 2019; Straneo et al., 2016). There are also few observational constraints on submarine melting, further complicating the effort to understand the ocean's forcing on glaciers. Instead the field has relied heavily on untested theory and models to estimate melt rates and their variability.

Submarine melting observations in Greenland are limited and, for the most part, indirect. While the total ice discharge is relatively well constrained at outlet glaciers Enderlin et al. (2014),King et al. (2018),Mankoff et al. (2019), partitioning this ice discharge into calving and submarine melting is an ongoing challenge. Melt rates have been derived from satellite data for floating ice tongues (Enderlin & Howat, 2013; Motyka



et al., 2011; Wilson et al., 2017), but these methods are not possible for the grounded or near-grounded termini that are most prevalent around Greenland. The total submarine meltwater flux can be estimated by measuring the oceanic transports of heat, salt, and mass through a fjord, called the flux-gate or fjord budgets method (Jackson & Straneo, 2016; Motyka et al., 2003, 2013). In Greenland, this method has been applied to synoptic ocean surveys (Inall et al., 2014; Rignot et al., 2010) and long-term moored records (Jackson & Straneo, 2016). However, this approach measures the total submarine meltwater from all upstream icebergs and glaciers and can have substantial errors from limits on the oceanographic record temporal and spatial resolutions (Jackson & Straneo, 2016). Finally, in Greenland, the terminus morphology has been measured with multibeam sonar and used to indirectly estimate or validate melt rates (Fried et al., 2015; Rignot et al., 2015). In Alaska, repeat sonar scans that track the terminus evolution in time—combined with records of ice velocity and calving—have provided a more direct estimate of melt rates (Sutherland et al., 2019), but these new methods have yet to be applied in Greenland.

Without direct measurements of submarine melting, our understanding of the ocean's impact on glaciers is currently limited by three main factors: (1) limited observations and knowledge of the controls on near-glacier ocean conditions in fjords, (2) lack of a validated theory for predicting melt rates based on near-glacier ocean conditions, and (3) unclear impact of submarine melting on calving and upstream glacier dynamics.

Submarine melting is expected to be a function of the near-glacier ocean conditions (temperature, salinity, circulation) and the subglacial discharge flux that forms from surface melt (Jenkins, 2011; Slater et al., 2015). Because of the valley morphology created by outlet glaciers and the large fluxes of surface meltwater delivered seasonally to the glacier bed, meltwater is likely routed to the base of the ice and down the main trunk of the glacier to the grounding line via a few relatively stable subglacial conduits (Fried et al., 2015; Lewis & Smith, 2009). At the terminus, this meltwater is ejected into the fjord at the glacier base and subsequently rises due to buoyancy, forming energetic plumes that entrain warmer ocean waters (Bendtsen et al., 2015; Mankoff et al., 2016; Motyka et al., 2003). Subglacial discharge drives enhanced turbulent transfer of ocean heat across the ocean-ice boundary layer, and thus, the overall rate of ocean heat delivery to the terminus will be function of the subglacial discharge flux and its distribution along the terminus (Carroll et al., 2015; Jenkins, 2011; Sciascia et al., 2013; Slater et al., 2015; Xu et al., 2012). Consequently, submarine melting is expected to be intimately coupled with atmospheric conditions, complicating the task of separating the oceanic versus atmospheric forcing of the GrIS.

The field has relied heavily on theory and models to derive melt rates as a function of the near-glacier ocean conditions and the subglacial discharge flux. A three-equation melt parameterization is typically used to calculate the terminus melt rate as a function of ice-adjacent ocean temperature, salinity and velocity, using empirical coefficients that have been derived from studies of sea ice and ice shelves (horizontal ice-ocean interfaces), and thus are not necessarily applicable to Greenlandic glaciers (Holland & Jenkins, 1999; Jenkins et al., 2010; McPhee et al., 1987; Straneo & Cenedese, 2015). Often this melt parameterization is coupled to a plume model from buoyant plume theory (Jenkins, 2011; MacAyeal, 1985), which is a well-developed theory to describe the evolution of a source of buoyancy (freshwater from subglacial discharge and submarine melt in this case) as it rises through a stratified fluid (Morton et al., 1956). The plume model estimates the ice-adjacent ocean temperature, salinity, and velocity ($\sim 1 \text{ m}$ from ice) based on the far-field ocean properties (>10-100 m from the ice) and the subglacial discharge flux. The ice-adjacent conditions are then plugged into the melt parameterization to calculate a melt rate. This coupled plume-melt theory has seen widespread use for estimating melt rates in several ways: on its own (Jenkins, 2011; Magorrian & Wells, 2016; Slater et al., 2016), in numerical ocean models (Carroll et al., 2017; Cowton et al., 2015; Kimura et al., 2014; Sciascia et al., 2013), and in observational studies that measure near-glacier ocean conditions (Jackson et al., 2017; Mankoff et al., 2016). In the last case, melt rates are derived using near-glacier observations, but the resulting melt rates should still be regarded as theoretical or modeled, not observationally validated.

Coupled plume-melt theory has been used to describe two regimes of melting across the terminus: (1) discharge-driven melting where vigorous plumes from subglacial discharge upwell along the terminus and (2) ambient melting away from discharge outlets. Because melting is expected to scale with ice-adjacent plume velocity (Holland & Jenkins, 1999; Jenkins, 2011), energetic discharge plumes should drive elevated melting, while ambient melt rates should be associated with less vigorous velocities and lower melt rates.



As a result, ambient melt is typically assumed to be negligible compared to discharge-driven melting, resulting in a focus on discharge-driven melting (Carroll et al., 2016; Cowton et al., 2015). This has led to recent progress in characterizing subglacial discharge plumes with observations from remotely operated kayaks and tagged seals (Everett et al., 2018; Jackson et al., 2017; Mankoff et al., 2016) and high-resolution ocean models of discharge plumes (Ezhova et al., 2018; Kimura et al., 2014; Xu et al., 2012).

Although models and theory indicate that ambient melt is negligible compared to discharge-driven melt, new observations suggest that ambient melt may play a significant role in the net terminus ablation (Slater et al., 2018; Sutherland et al., 2019; Wagner et al., 2019). If buoyancy from submarine melt and subglacial discharge is the only driver of ocean velocity along the terminus, then the ambient melt regime will be characterized by weak plumes that rise over a short distance (on the order of 1-10 m) and then intrude into the fjord in a series of stacked layers (Magorrian & Wells, 2016). However, this framework neglects the realistic 3-D structure of ocean circulation along the terminus. Slater et al. (2018) combine an ocean model with observed near-glacier velocity to show that horizontal recirculations from the discharge plume will enhance ocean velocity over large swaths of the terminus and likely elevate ambient melt rates, which is corroborated by the surface morphology and ice flux budget in Wagner et al. (2019). At LeConte Glacier in Alaska, direct estimates of submarine melting from multibeam sonar (Sutherland et al., 2019) and near-glacier surveying by autonomous kayak Jackson et al., (2019) reveal that ambient melt rates may be 100× higher than expected from standard plume-melt theory—likely due to a combination of the horizontal velocity field (Slater et al., 2018) and errors in the empirical coefficients that are critically embedded in the melt parameterizations. Though the magnitude of ambient melt at Greenlandic glaciers has yet to be directly measured, these studies suggest that both submarine melt regimes (discharge-driven and ambient melt) are critical to the total terminus ablation. The growing discrepancy between melt theory and observations highlights a need for new observations at the ice-ocean boundary to develop an updated, generalizable theory for submarine melting.

The intertwined dynamics of melting and subglacial discharge might play an important role in understanding where and how submarine melting can effect calving. Subglacial plumes can reach the surface—if the grounding line is relatively shallow, discharge is sufficiently large, and/or if fjord stratification is weak—and can be visible as a vigorous, sediment-laden plume in satellite images (Fried et al., 2015; Jackson et al., 2017; Mankoff et al., 2016). While lower discharge plumes may not rise as high in the water column, the impact of buoyant plumes on terminus melt can be large even for small discharge plumes (Carroll et al., 2016), particularly if sufficient melting near the grounding line generates an overhanging geometry that is favorable to calving via sloughing (Fried et al., 2015; Fried et al., 2019; Slater et al., 2017). In fact, submarine melt and subaerial calving of overlying ice in the vicinity of subglacial conduits are the dominant mechanisms driving seasonal terminus change for thin, well-grounded outlet glaciers in west Greenland (Chauché et al., 2014; Fried et al., 2018; Luckman et al., 2015), and possibly around the ice sheet. This causes thin glaciers to form local terminus embayments in the vicinity of subglacial conduits and often visible surface sediment plumes (Fried et al., 2018). Thicker glaciers likely also experience submarine melt and terminus undercutting near subglacial conduits, but the thick ice possibly insulates against downward surface crevasse propagation, and as a result, the seasonal retreat pattern is more uniform across the glacier width (Fried et al., 2018).

5.3. Atmospheric Controls on Outlet Glaciers

The net ice sheet surface mass balance is primarily the result of an imbalance between winter snowfall accumulation and summer surface melt. Although Greenland air temperatures have been increasing since the mid-20th century (Trusel et al., 2018), mean summer temperatures across the GrIS became positive in the mid-1990s (Mernild et al., 2011; van den Broeke & Lenaerts, 2014). In response to this combined natural and anthropogenic atmospheric warming, GrIS surface melt has increased substantially (Fettweis et al., 2017; Noël et al., 2017; van den Broeke et al., 2016), resulting in a decreasing trend in SMB of 10.2 Gt yr⁻¹ during 1991–2015 (van den Broeke et al., 2016). However, because the atmospheric state can change dramatically over annual and shorter timescale and because Greenland SMB is highly sensitive to atmospheric conditions, variations in SMB are significant (Box, 2013). For example, the long-term trends in atmospheric warming and particularly strong GBI combined in 2012, resulting in GrIS meltwater runoff that was unprecedented within at least the last 350 years (Trusel et al., 2018).

Both long- and short-term variations in SMB can have appreciable impacts on ice flow. Just like their land-terminating counterparts, outlet glaciers are susceptible to surface climate changes that impact surface



slopes: Surface melt causes surface elevation lowering that is enhanced at lower elevations, where temperatures are warmer thus steepening the outlet glacier surface (Felikson et al., 2017; Noël et al., 2016). This can create a positive feedback that enhances ice flow into warmer, low-elevation regions and further exacerbates differences in melt and slope. Changes in outlet glacier surface slopes from along-glacier surface melt are not well investigated, although rates of surface melting at the termini of some outlet glaciers exceed 1 m/year water-equivalent (Noël et al., 2016). The influence of surface melt on surface slope is the primary means by which climate influences ice dynamics in earth system models, not necessarily because it has the largest influence on dynamics, but because its effects on ice flow are well understood.

In addition to SMB-linked slope changes, the generation and variability of surface melt has importance for multiple processes operating on outlet glaciers including: (1) subglacial discharge, which drives submarine terminus melt, calving, and fjord circulation (discussed above); (2) the spatio-temporal distribution of sub-glacial water pressures, which influence friction at the ice-bed interface; and (3) changes in ice rheology from release of latent heat due to refreezing of meltwater within voids, like crevasses and moulins, in the ice sheet. These processes influence outlet glacier behavior over a wide range of time and space scales, demonstrating the important role of surface climate forcing on the future evolution of outlet glaciers. There has been considerable work done to build regional GrIS climate models (Fettweis et al., 2017; Noël et al., 2017), and these models are well-validated where observations permit (Vernon et al., 2013). Validation sites are limited, however, and there is currently no way to validate the time-varying discharge of subglacial water that runs off the GrIS into the ocean. Research addressing this question for subglacial runoff in land-terminating areas suggests variable model success in capturing observations (Smith et al., 2017)

Around the GrIS margins, where melt exceeds local storage capacity, the vast majority of surface meltwater is efficiently routed to the ice-bed interface through moulins (Catania & Neumann, 2010; Das et al., 2008; Smith et al., 2017). The spring onset of meltwater flux to the bed raises subglacial water pressure at the ice-bed interface because of limited water storage capacity and decreases friction at the bed, facilitating faster ice flow (Andrews et al., 2014; Iken et al., 2017; Meierbachtol et al., 2013). Such observations are taken from land-terminating parts of the ice sheet using boreholes to suggest that the subglacial drainage system (and thus the water pressure distribution at the bed) is complex and composed of both subglacial conduits that are sensitive to creep closure and unchannelized bed regions that gain efficiency as the melt season progresses (Andrews et al., 2014; Bougamont et al., 2014; Hoffman et al., 2016). Observations suggest that there is good agreement between regionally modeled ice sheet runoff and ice flow speed in outlet glaciers (Moon et al., 2014; Rathmann et al., 2017), but it is not clear if this is a direct relationship because runoff can alter subglacial hydrology as well as enhance upwelling, melt, and retreat at the terminus (Fried et al., 2018); both of which will influence ice flow resistance. A few studies have examined the influence of subglacial conditions on outlet glacier ice flow speeds using surface and/or borehole geophysics, finding that a highly deformable ice layer exists near the base of each glacier, which is interpreted to be temperate pre-Holocene ice that contributes substantially to total ice motion (Hofstede et al., 2018; Lüthi et al., 2002). Further, outlet glaciers are likely to be underlain with subglacial sediments (Christianson et al., 2014; Hofstede et al., 2018; Walter et al., 2014), with complex spatial patterns of basal slipperiness and suggesting that basal conditions for outlet glaciers may vary over small (~100 m) scales. Finally, because glaciers terminate in the ocean with water depths of 10s-100s of meters, high water pressures exist year round and seasonal changes in flow at glacier termini can result in ice thickness changes that are felt far inland (Young et al., 2019).

Although the majority of surface meltwater routes efficiently to moulins (Smith et al., 2017), there is some evidence for meltwater retention within or beneath the ice sheet. If englacial water refreezes, it releases latent heat into the surrounding ice in a process termed cryo-hydrologic warming (Phillips et al., 2010). Indeed, temperature profiles in boreholes reveal warmer englacial temperatures than would be present from heat diffusion and dissipation alone (Harrington et al., 2015; Lüthi et al., 2015). This heat has the ability to soften the ice, which has a temperature-dependent rheology and may be responsible for some of the observed enhanced ice flow in Greenland (Poinar et al., 2016; van der Veen et al., 2011). Where fast speeds reduce the transit time for ice through the ablation zone (where surface melt is generated), however, the potential impact of cryo-hydrologic warming is likely small (Poinar et al., 2016). In addition to ice temperature and water content, ice viscosity is also a function of the effective strain rate. Bondzio et al. (2017) show that sustained Jakobshavn Isbrae acceleration, following the termination of a retreat period, is due to a decrease in ice viscosity by four orders of magnitude. Because of low basal drag for Jakobshavn Isbrae, most of the driving stress is supported by the lateral shear margins (Shapero et al., 2016; van der Veen et al., 2011), and







sudden changes in the calving front induce stress perturbations that promote an instantaneous drop in ice viscosity within the shear margins, which further weaken via strain heating (Bondzio et al., 2017). Such observations suggest that ice sheet thermal processes and ice sheet viscosity changes are critical to model to understand changing outlet glaciers.

6. Impacts of a Changing GrIS to the Arctic

While mountain glaciers and ice caps were the largest land ice contributor to sea level rise during the 1990s and 2000s (compared to Greenland and Antarctica), the GrIS began to match and exceed that contribution around 2013 (Chen et al., 2017). The much larger reservoir of potential sea-level rise contained in the GrIS (7.42±0.05 m (Morlighem et al., 2017)), versus mountain glaciers and ice caps (0.32±0.08 m (Farinotti et al., 2019)), gives Greenland the capacity to maintain current (or higher) loss rates much further into the future. This underlies the importance of understanding the mechanisms responsible for mass change in order to improve estimates of the rates of change that are possible. The projected range of future GrIS mass loss across different greenhouse gas emissions scenarios (Vaughan & Comiso, 2014) shows strong overlap over the next several decades, but diverges substantially over the next several centuries (Aschwanden et al., 2019; Clark et al., 2016; Mengel et al., 2016). Projections of future GrIS mass loss under an RCP8.5 forcing (a representative concentration pathway producing 8.5 W/m² radiative forcing in 2100 (Vaughan & Comiso, 2014)) suggest that GrIS mass loss could reach up to 33 cm by 2100 (Aschwanden et al., 2019). Between 1972 and 2018, the

ice dynamics contribution to total mass loss was $66\pm8\%$ (9.1 mm), and SMB was $34\pm8\%$ (4.6 mm) (Mouginot et al., 2019); however, as air temperatures continue to increase, large surface melt events (Nghiem et al., 2012) will occur with increased frequency (Trusel et al., 2018), and SMB losses are expected to increase (Aschwanden et al., 2019; Fyke et al., 2014). Unfortunately, partitioning future mass loss from Greenland is still difficult owing to uncertainties in surface melt processes (Aschwanden et al., 2019; Fürst et al., 2015) and the complexities of interconnected mechanisms described above.

The impacts of GrIS mass loss not only depend on the magnitude of mass loss, but its spatial distribution as well. Local sea-level change experienced in coastal cities is sensitive to the "fingerprints" of mass loss from the ice sheets—the amount of SLR expected across the globe depends on where the mass loss is concentrated (Larour et al., 2017; Mitrovica et al., 2009; Mitrovica et al., 2018). For example, in the United States, Los Angeles will experience greater sea-level rise due to Greenland mass loss than New York, and most U.S. locations are more sensitive to eastern Greenland mass loss than western Greenland mass loss (Larour et al., 2017).

Spatial variations in the delivery of freshwater (and sediments) to the ocean via outlet glaciers also affects ocean dynamics and nutrient fluxes (Figure 4). This freshwater influences ocean stratification and circulation, both locally within fjords and regionally around the continental shelves of Greenland. Freshwater input from glaciers—from subglacial discharge and submarine melting of glaciers and icebergs—can drive a fjord-wide exchange flow (Carroll et al., 2015; Sciascia et al., 2013), which may impart a positive feedback on future ice mass loss from Greenland by enhancing ocean heat delivery through fjords (de la Guardia et al., 2015). Additionally, these sources of freshwater enter the ocean at depth (often many 100s of meters below the surface) and drive vigorous upwelling of deep waters. Thus, Greenland outlet glaciers impact the ocean both as a source of a freshwater and as a powerful mechanism for vertically redistributing oceanic waters masses, pumping heat and salt from the deep ocean to the surface layer (Beaird et al., 2015; Beaird et al., 2018).

The partitioning of mass loss between solid and liquid components is also important, as the timing of iceberg melt is offset from other sources (e.g., peaking in fall rather than summer) and iceberg melt injects freshwater



into the surface of the ocean over much broader regions (Enderlin et al., 2018; Moon et al., 2017). The ratios of solid to liquid freshwater flux vary substantially around the ice sheet (Enderlin et al., 2014; Mouginot et al., 2019), with iceberg freshwater fluxes dominant over all other terms in fjords connected to high discharge outlet glaciers (Enderlin et al., 2016; Moon et al., 2017). At the ice sheet scale, however, liquid freshwater fluxes from GrIS meltwater runoff dominate solid fluxes (\sim 1300 to \sim 1170 km³ yr⁻¹) with small liquid fluxes from Greenland's peripheral glaciers and ice caps (\sim 920 km³ year⁻¹) and terrestrial runoff (\sim 860 km³ year⁻¹) since 2010 (Bamber et al., 2018). Variations in the magnitudes of these freshwater flux components impact the circulation pathways of the freshwater in the Arctic (Böning et al., 2016; Luo et al., 2016), with the capacity to impact deep water formation in the North Atlantic and globally (Böning et al., 2016).

Outlet glaciers are also important for producing and redistributing nutrients to the Arctic ocean (Cape et al., 2018; Hopwood et al., 2018; Kanna et al., 2018; Meire et al., 2015; Meire et al., 2017; Overeem et al., 2017). There are two primary mechanisms at play: (1) direct nutrient input to the grounding line via subglacial discharge that supplies turbid melt water and (2) enhanced nutrient upwelling from the deep ocean due to the input of subglacial discharge at depth. Research at multiple marine-terminating glaciers around Greenland suggests the second mechanism is more important, with upwelling nitrate fluxes that are 10 times larger than discharge fluxes (Cape et al., 2018; Hopwood et al., 2018; Kanna et al., 2018), and direct comparison of fjord productivity near marine-terminating versus land-terminating glaciers confirms the importance of deep water input from marine-terminating glaciers for sustaining high productivity (Meire et al., 2017). In addition, the surface layer in fjord systems is undersaturated in CO_2 due to the input of glacial meltwater, indicating that fjords are strong CO₂ sinks (Meire et al., 2015) and thus important, but overlooked, components of the Arctic carbon cycle. As outlet glaciers continue to retreat into the interior of the GrIS, many will retreat up subglacial beds that rise steeply above sea level behind the present-day terminus (Catania et al., 2018), effectively eliminating their connection to the ocean. This decoupling of glaciers (and their meltwater) from the ocean will have important consequences for the timing and spatial distribution of nutrient delivery and the carbon cycle in the Arctic ocean (Hopwood et al., 2018; Meire et al., 2015).

7. Future Research Needs

At the heart of improved predictions of outlet glacier contributions to sea level and freshwater forcing of neighboring ocean basins is our ability to accurately model outlet glacier dynamic changes and how these changes influence, and are influenced by, climate. In turn, this requires improved understanding of the processes that influence glacier dynamics and concerted effort to link ice sheet models to existing ocean and atmosphere models. Importantly, these links need to be two way because of the intimate interplay of processes at work driving change at the ice sheet margin, and the ability of this change to, in turn, impact the global climate system (Golledge et al., 2019). Thus, future research advances require more observations of critical processes that govern glacier dynamics, better integration of critical processes (e.g., terminus change) in numerical ice flow models based on these observations, and continued development of, and collaboration within, an interdisciplinary and diverse scientific community.

7.1. Meeting Observational Needs

Spaced-based observations of Earth's glaciers and ice sheets have been acquired at an accelerated pace over the last few decades as new satellite missions are launched (e.g., ICESat and GRACE), existing ones are prolonged (e.g., Landsat, ICESat-2, GRACE-FO, CryoSat-2, Sentinel), and nongovernmental agencies have become involved in polar imaging (e.g., DigitalGlobe). These instruments provide observations of many of the essential variables involved in glacier dynamic changes including surface elevation, terminus position, and surface speed. The timescales of data acquisition have also improved with many observations made at month-to-seasonal frequency, which is necessary to remove observation biases. Surface speeds and elevations are now automatically generated for the entire ice sheet and provided to public databases with regular updates (Fahnestock et al., 2016; Howat, 2017; Howat et al., 2014; Joughin et al., 2010; Rosenau et al., 2015). Care must be taken to ensure long-term continuation of these important observations, particularly for studies that examine the attribution of ice sheet change to anthropogenic and/or natural climate change, as has been done for alpine glaciers worldwide (Marzeion et al., 2014). The extraction of terminus positions from satellite data, on the other hand, remains a manual process performed as needed by individual researchers. Machine learning techniques are beginning to make this task less onerous (Lea, 2018; Mohajerani et al., 2018), although terminus mapping in the often cloudy, ice-congested, and seasonally shadowed conditions may still require semimanual delineation by a well-trained operator, a particularly time-consuming endeavor. We encourage community collaboration to rapidly make these records available for the satellite data archive and continued efforts focused on automated picking of termini. These data will be valuable for multiple reasons, but importantly by identifying terminus "hot spots" over the satellite era (regions of focused, rapid change), we can focus ice sheet models to produce higher resolution solutions for individual glacier basins and use these estimates to update sea-level predictions.

Important observational gaps also exist in terms of understanding how subglacial processes influence glacier dynamics, noted by the lack of an agreed-upon sliding law (Joughin et al., 2019). While additional in-situ geophysical observations are needed (e.g., drilling, passive and active seismic, etc.), such observations are costly and logistically difficult on heavily crevassed outlet glaciers. Further, while models have rapidly accelerated in development to the point where now they can reproduce the bulk of some of the hard-won observations (Hoffman et al., 2016; Stevens et al., 2018; Werder et al., 2013), many of these observations take place on land-terminating portions of the ice sheet. Evidence from ground-based geophysics on a small number fast-moving outlet glaciers points to a sediment layer on the order of 100 m thick that underlies with highly variable slipperiness (Christianson et al., 2014; Hofstede et al., 2018; Walter et al., 2014). This suggests a need for additional ground-based observations aimed at characterizing the possible range of subglacial conditions that exist under outlet glaciers. Further, since subglacial erosion and deposition contribute to changes in subglacial water pressure (Andrews et al., 2014; Catania & Paola, 2001), and influence glacier stability (Alley et al., 2007; Brinkerhoff et al., 2017; Nick et al., 2007), the rates of these processes need to be measured as well. Additionally, understanding how subglacial drainage evolves over time and how this, in turn, influences outlet glacier velocity is still uncertain and ripe for additional model development. In particular, the redistribution of stresses within an outlet glacier are likely to adjust on a range of time scales in response to changing subglacial conditions. Estimates of glacier stresses minimally require time-evolving observations of surface speed and elevation. Sufficiently highly resolved surface topographic observations are available at least annually (Porter et al., 2018); however, shorter time scale changes may also be relevant (Enderlin, Carrigan, et al., 2018). In addition, it is unclear how basal resistive stresses behave outside of the regions covered with subglacial conduits, arguing for increased spatial resolution in observational data needed for force balance studies.

In addition to its influence on basal properties, surface melt plays a critical role in controlling outlet glacier dynamics through its influence on submarine terminus melt hence possibly the rate of calving. Unfortunately, we currently lack the ability to regularly and uniformly validate subglacial discharge estimates in the submarine environment. Most estimates come from regional climate model evaluations of surface melt-derived runoff with routing via the subglacial drainage system. Partitioning of runoff within a given drainage system introduces errors in runoff estimates, and some meltwater could be stored subglacially or in the surface firn layer (Forster et al., 2014; Rennermalm et al., 2013). Seismic techniques have been proven viable to elucidate patterns in subglacial water movement and storage (Zhan, 2019) as well as subglacial discharge flux through time (Bartholomaus et al., 2015), and we argue for widespread observations of sub-glacial tremor in Greenland in order to reduce uncertainty in model-based estimates of subglacial water storage, transport, and discharge.

The ocean influence on the GrIS has recently been a target for accelerated research with three emerging research needs related to ice-ocean interactions. The first need involves understanding the ocean conditions near glacier termini, how they vary in time, and what drives this variability. This involves understanding the drivers of ocean circulation at the fjord scale and fjord-shelf exchange, as well as larger scale ocean processes (the origin of AW and PW, decadal variability, etc.). While there has been considerable progress on this over the last few decades, we critically lack systematic, long-term monitoring of ocean properties around the GrIS (Straneo et al., 2019). This should include observations of heat content and fjord/shelf conditions with an aim to improve understanding of how fjord and shelf properties are related to near-glacier properties and then how those are related to melt rates at glacier termini. In addition, we prioritize velocity data within fjords because these data are critical for understanding fjord-scale processes, measuring freshwater export, and monitoring heat transport. Many groups have conducted difficult and expensive observational campaigns to obtain moored observations in select glacier fjords and around the GrIS (Carroll et al., 2018; Gladish et al., 2015; Gladish et al., 2015; Mortensen et al., 2014; Straneo et al., 2016), but these efforts only began within the past ~10 years and are often not maintained long term, so there remains a limited understanding of the temporal evolution in ocean forcing of the ice sheet. There is significant opportunity for coordinated

ocean-observing platforms as suggested by Straneo et al. (2019) to overcome these observational bottlenecks and provide critical data necessary to address this need.

The second research priority related to ice-ocean interactions involves improved understanding of terminus submarine melt rates and how they relate to ocean conditions near the terminus. There has been rapid advancement in obtaining observations close to glacier termini via autonomous platforms (Jackson et al., 2017; Mankoff et al., 2016), but these have been limited to a few glaciers and do not allow for direct measurements of melting nor validation of modeled melt rates. In addition, the existing near-glacier observations are taken in summer, when glacier termini are easier to access by boats; however, biases in the spatial and/or temporal coverage of observations may limit our understanding of the seasonal evolution of melt processes. Finally, recent studies (Carroll et al., 2016; Slater et al., 2018; Sutherland et al., 2019) suggest that melt rates might be significantly underestimated by the prevalent melt parameterizations (Holland & Jenkins, 1999; Jenkins, 2011; Magorrian & Wells, 2016). Thus, we argue for more observations—near the ocean-ice boundary and across a range of glacier-fjord systems—to obtain a generalized, observationally validated theory that can link terminus melt rates to ocean conditions and subglacial discharge flux.

The third research need for ice-ocean interactions involves improving our understanding of how submarine melt impacts the glacier, including calving. Observations of melt-induced calving primarily come from smaller glaciers that do not have iceberg-congested fjords, biasing our understanding of calving processes to these glaciers. We have no estimates for how important submarine melt is for large-flux, deeply grounded outlet glaciers with persistent, year-round mélange. In particular for these larger glaciers, the importance of submarine melt on the propagation of basal crevasses may be important to determining the calving flux (James et al., 2014; Murray, Selmes, et al., 2015).

A final observational need includes improved estimates of fjord bathymetry and basal topography in terminus regions of outlet glaciers. These data are of fundamental importance to glacier stability, as well as to understanding ocean heat transport to the ice sheets and freshwater exchange with the Arctic Ocean. Despite the rapid acquisition of new observations from projects like oceans-melting-greenland in many fjords (Morlighem et al., 2016), there remain numerous glaciers where topography is poorly constrained, primarily those along Northern Greenland (Morlighem et al., 2017). Many of these fjords, as well as those along SE Greenland, are choked with icebergs year-long, making it difficult to gain ship access close to the glacier termini. Further, radar-based efforts to map the ice sheets are to be commended for the rapid pace at which observations have accelerated; however, the heavily crevassed, meltwater-rich, deep outlet glaciers remain a challenge for radar-energy propagation to the bed. Observations of basal topography for a few key outlet glaciers should be prioritized, possibly through drone, or helicopter support.

7.2. Including Critical Processes in Models

Over the last several years, as the sensitivity of GrIS outlet glaciers to changes in atmospheric and oceanic forcing has become more apparent, there have been tremendous advances in our understanding of critical processes controlling ice dynamics as well as in numerical modeling of glaciers and ice sheets. However, progress in observations and models has largely occurred side-by-side, with a few exceptions where observations for a specific glacier system are sufficiently dense that observation-based numerical modeling of the coupled ice-atmosphere-ocean system can be performed (Bondzio et al., 2017; Seroussi et al., 2014; Todd et al., 2018). As a result, we argue for model development to prioritize the inclusion of data via assimilation techniques (Goldberg et al., 2015; Heimbach et al., 2019) to produce ice sheet projections that are fully consistent with past observations. In part, this remains a challenge because of the lack of observations of changing bed and terminus conditions, but also because of the possibly high dependence of ice dynamics to unmeasured quantities (e.g., small-scale bathymetry or bed characteristics, factors influencing ice rheology). Further, this requires an assurance that data—including processes study results—are findable, accessible, interoperable, and reusable (FAIR) (Wilkinson et al., 2016). This includes efforts to establish and use standard data formats and to create suites of data that are curated to tackle interdisciplinary science questions. Well-designed data access and visualization systems are also invaluable for reducing startup times for new research and researchers, facilitating connections between glaciology and other disciplines and providing tools and information for connecting with stakeholders.

The shift in focus to more realistic boundary processes in numerical ice flow models should result in the development of parameterizations that can be feasibly implemented into large-scale numerical models (i.e., that have reasonable computational demands). This task is challenging for several reasons. First, there is

still debate on how to parameterize important processes like submarine terminus melt, calving, and basal sliding. The recent surge of new observational data around the GrIS has enabled observation-based testing of a number of existing process parameterizations that control glacier flow (Bondzio et al., 2017; Choi et al., 2018; Enderlin & Bartholomaus, 2019; Stearns & van der Veen, 2018), but these must be generalized across all glacier systems and validated with observations. Second, improvements in model resolution and nested modeling techniques are beginning to allow full ice sheet models to resolve outlet glaciers (Aschwanden et al., 2016; Aschwanden et al., 2019), but resolution challenges continue. On the oceanic side, connecting ocean processes across a range of scales, from the continental shelf to the turbulent ice-ocean boundary layer, remains an open challenge to the research community. Small-scale plume models (~1 m resolution) can resolve the upwelling plumes at glacier termini; fjord models (~100 m resolution) capture larger circulation patterns between the glacier and shelf ocean, and regional models (~1 km resolution) reveal shelf and subpolar gyre dynamics. Typically, these scales are studied separately, but going forward these various scales must be integrated more effectively in order to model the interaction between the large-scale ocean and the GrIS.

Third, several of the processes important to ice-ocean interactions require coupling to atmospheric parameters, and integration across all of these models is challenging, particularly given the observed sensitivity of the GrIS to atmosphere and ocean change and the substantial uncertainties in future atmospheric and oceanic conditions (Pattyn et al., 2018). This is hard and costly, but remains a necessary step so that ocean models no longer ignore solid ice discharge (Luo et al., 2016), convert solid ice discharge immediately to liquid flux, and/or input all freshwater into the ocean surface (Böning et al., 2016; Gillard et al., 2016) but instead incorporate time-evolving partitioned freshwater fluxes appropriately. Simpler modeling approaches could, and have been, applied to project glacier or changes in glacier boundary conditions over shorter timescales. For example, Bamber et al. (2018) use the relationship between SMB and ice discharge estimates to extrapolate discharge to years without observations. Slater et al. (2019) use relationships between observed terminus change and submarine melt estimates, forced using future atmospheric change scenarios, to predict 21st century GrIS terminus change. Although there is utility in these methods, they are unlikely to yield accurate projections of GrIS mass loss over long time scales since the evolution of the glacier geometry as it recedes will undoubtedly influence its sensitivity to climate change and ice discharge. Thus, despite the difficulties coupling numerical models and uncertainties in these models, they are the most likely to yield accurate mass loss estimates and we recommend that future modeling efforts focus on their continued development.

7.3. Community Coordination and Capacity Building

Solutions to many of the research challenges presented above are within reach, given sustained, coordinated funding. More frequent in-situ observations of deep ocean circulation around the GrIS coupled with long-term observations within multiple glacial fjords would provide much-needed data to test parameterizations of ice-ocean exchanges. Thus, there is significant opportunity for funding agencies to work across internal boundaries and with international partners to facilitate the development and maintenance of such observations. In lieu of large funding schemes for collaborative observational platforms, we urge the dismantling of boundaries that hamper intellectual collaboration. The recent rapid growth of interest in outlet glaciers in Greenland, and ice-ocean interactions in particular, has been matched with rapid expansion of knowledge on processes and information that would have taken the previous generation of glaciologists much longer to acquire in isolation. We argue for an even more expansive community of scientists interested in the GrIS because the inclusion of outside knowledge and experience drives innovation in our field. This includes increasing diversity in our field, expanding international collaborations (including with indigenous knowledge holders and local communities) and integrating across an even wider range of scientific disciplines. These necessary advances can, in part, be accomplished through joint AGU Fall Meeting sessions across typically separated focus groups, special conferences focused on specific problems that remain unresolved at the boundaries of traditional disciplines, greater international cooperation similar to the joint U.S.-U.K. efforts underway at Thwaites Glacier funded through coordinated efforts at NERC and NSF, and a wider range of mechanisms that engage and support new researchers within our community. Achieving these goals will also advance opportunities to explore the impacts of the current rapid reconfiguration of the GrIS in a system science context, where many of today's most exciting discoveries are being made. In addition, because so many of the recent discoveries in ice-ocean exchanges in Greenland have occurred through



teams composed of diverse scientists, we suggest that awards from scientific societies be augmented to recognize the enormous team efforts needed to produce quality observations in many of the far-flung reaches of the poles. AGU could lead such an effort.

8. Outlook

As a community, glaciology has moved quite rapidly in the last few decades spurred by outside interest, increased quantity, quality, and access to data, and improved methods for interpreting and reproducing observations with ice flow models. However, we still have not solved the problem of how much sea level will rise, when, and where it will come from. Much work is focused on these goals, and we encourage greater coordination from funding agencies, data providers, and within research communities to make rapid progress on this grand challenge. Observational challenges remain, primarily in the ice-ocean and at the ice-bed interfaces, but these can be overcome with targeted, well-coordinated, long-term monitoring and improved technology for observing bed properties and submarine termini. These efforts should be prioritized by funding agencies. Model development is also making great leaps in resolution and incorporation of processes. That community has come a long way from the Fourth Assessment Report from the IPCC (Lemke et al., 2007), which stated that although observations indicated the importance of dynamical processes in ice sheets, they were not included in ice sheet models because of the lack of consensus on their ability to increase ice sheet loss in a warming climate. Further progress is needed to link ice sheet models to the climate and ocean models that exist outside of our community. This is particularly important for ice-ocean exchanges because of the range of impacts that are associated with a melting GrIS.

Glossary

- ice sheet mass balance The difference between ice sheet mass accumulation (e.g., through snowfall) and ablation (e.g., through meltwater runoff and calving). Ice sheet mass balance is usually measured on an annual timescale to capture seasons of accumulation and ablation.
- **surface mass balance** The difference in mass caused only by changes in accumulation and ablation independent of calving and iceberg discharge (which are associated with ice dynamics). Primary contributors to surface mass balance are snowfall and meltwater runoff.
- **ice dynamics** Describes the motion within large bodies of ice, such as glaciers and ice sheets, subject to numerous processes that influence the rheology of ice and the boundary conditions.
- **calving** The process of breaking ice from the end of a marine-terminating glacier or ice shelf, adding broken ice to the ocean, which are then referred to as icebergs (bergy bits and growlers are smaller than icebergs).
- **grounding line** The subglacial location at which the glacier terminus region loses contact with ground beneath it. Sometimes also referred to as a grounding zone. Up-glacier of the grounding line, the glacier is 'grounded' while down-glacier the glacier is floating.
- **moraine** Unstratified and unsorted rock and sediment piles moved into place via sediment deposition, glacier erosion and flow. Moraines can exist in front of, next to, and along interior flowlines of a glacier. Oceanography disciplines use the term sills to describe moraines.
- **tidewater glacier cycle** The idea that tidewater glaciers undergo cycles of dynamic behavior that oscillate between slow advance and rapid retreat, punctuated by periods of stability. This oscillation occurs with little sensitivity to climate forcing.
- **moulins** A tubular chute through which water enters the glacier from the surface. Moulins may be formed by surface meltwater entering a crevasse, with water flow maintaining the channel.
- **firn** As surface snow remains past one year (one melt season) it is compressed, becoming more dense than snow but less dense than glacier ice. This intermediate material is called firn. As firn densifies the pore spaces within it becomes less connected and ice crystal size usually increases.
- **glacier hypsometry** Hypsometry describes the distribution of glacier surface elevation with respect to sea level within an area of interest, with positive values being above sea level and negative values below sea level.



Acronyms

- SMB Surface Mass Balance
- GrIS Greenland Ice Sheet
- NAO North Atlantic Oscillation
- GBI Greenland Blocking Index
- AMO Atlantic Multidecadal Oscillation
 - AW Atlantic Water
 - **PW** Polar Water
- PARCA Program for Arctic Regional Climate Assessment
 - **ATM** Airborne Topographic Mapper
- ICESat Ice, Cloud, and land Elevation Satellite
- **GRACE** Gravity Recovery And Climate Experiment
 - LIA Little Ice Age
 - RCP Representative Concentration Pathway
 - **IPCC** Intergovernmental Programme on Climate Change
 - FAIR Findable, Accessible, Interoperable, and Reusable
 - NSF National Science Foundation
 - NERC Natural Environment Research Council

Acknowledgments

We thank Bryn Hubbard for providing the opportunity and guidance for this review. We would like to acknowledge funding from NASA and NSF to to all of the manuscripts authors over their careers as well as the efforts of their students, postdocs, and collaborators who have contributed indirectly to the knowledge in this review. We also thank Mathieu Morlighem for critical comments that improved the manuscript prior to submission. Finally, we appreciate the thorough and thoughtful reviews by three anonymous reviewers for providing critical suggestions that improved the overall manuscript. No data are presented in this manuscript.

References

- Åkesson, H., Nisancioglu, K. H., & Nick, F. M. (2018). Impact of Fjord Geometry on Grounding Line Stability. Frontiers in Earth Science, 6(71), 71. https://doi.org/10.3389/feart.2018.00071
- Alley, R. B., Anandakrishnan, S., Dupont, T. K., Parizek, B. R., & Pollard, D. (2007). Effect of Sedimentation on Ice-Sheet Grounding-Line Stability. Science, 315(5), 1838–1841. https://doi.org/10.1126/science.1138396
- Amundson, J. M., & Burton, J. C. (2018). Quasi-Static Granular Flow of Ice Mélange. Journal of Geophysical Research: Earth Surface, 123, 2243–2257. https://doi.org/10.1029/2018JF004685
- Amundson, J. M., Fahnestock, M. A., Truffer, M., Brown, J., Brown, J., Lüthi, M. P., & Motyka, R. J. (2010). Ice mélange dynamics and implications for terminus stability, Jakobshavn Isbræ, Greenland. *Journal of Geophysical Research*, 115, F01005. https://doi.org/10.1029/ 2009JF001405
- Andrews, L. C., Catania, G. A., Hoffman, M. J., Gulley, J. D., Lüthi, M. P., Ryser, C., et al. (2014). Direct observations of evolving subglacial drainage beneath the Greenland Ice Sheet. Nature, 514(7520), 80–83. https://doi.org/10.1038/nature13796
- Aschwanden, A., Fahnestock, M. A., & Truffer, M. (2016). Complex Greenland outlet glacier flow captured. *Nature Communications*, 7(1), 1–8. https://doi.org/10.1038/ncomms10524
- Aschwanden, A., Fahnestock, M. A., Truffer, M., Brinkerhoff, D. J., Hock, R., Khroulev, C., et al. (2019). Contribution of the Greenland Ice Sheet to sea level over the next millennium. *Science Advances*, 5(6). https://doi.org/10.1126/sciadv.aav9396
- Bamber, J. L., Griggs, J. A., Hurkmans, R. T. W. L., Dowdeswell, J. A., Gogineni, S. P., Howat, I. M., et al. (2013). A new bed elevation dataset for Greenland. *The Cryosphere*, 7(2), 499–510. https://doi.org/10.5194/tc-7-499-2013
- Bamber, J. L., Tedstone, A. J., King, M. D., Howat, I. M., Enderlin, E. M., van den Broeke, M. R., & Noël, B. P. Y. (2018). Land Ice Freshwater Budget of the Arctic and North Atlantic Oceans: 1 Data, Methods, and Results. *Journal of Geophysical Research: Oceans*, 44, 1827–1837. https://doi.org/10.1002/2017JC013605
- Bartholomaus, T. C., Amundson, J. M., Walter, J. I., O'Neel, S., West, M. E., & Larsen, C. F. (2015). Subglacial discharge at tidewater glaciers revealed by seismic tremor. *Geophysical Research Letters*, 42, 6391–6398. https://doi.org/10.1002/(ISSN)1944-8007
- Bartholomaus, T. C., Anderson, R. S., & Anderson, S. P. (2008). Response of glacier basal motion to transient water storage. *Nature Geoscience*, 1(1), 33–37. https://doi.org/10.1038/ngeo.2007.52
- Bartholomaus, T. C., Larsen, C. F., & O'Neel, S. (2013). Does calving matter? Evidence for significant submarine melt. Earth and Planetary Science Letters, 380, 21–30. https://doi.org/10.1016/j.epsl.2013.08.014
- Bartholomaus, T. C., Larsen, C. F., O'Neel, S., & West, M. E. (2012). Calving seismicity from icebergsea surface interactions. Journal of Geophysical Research, 117, F04029. https://doi.org/10.1029/2012JF002513
- Bartholomew, I. D., Nienow, P. W., Sole, A. J., Mair, D. W. F., Cowton, T. R., King, M. A., & Palmer, S. J. (2011). Seasonal variations in Greenland Ice Sheet motion: Inland extent and behaviour at higher elevations. *Earth and Planetary Science Letters*, 307(3-4), 271–278. https://doi.org/10.1016/j.epsl.2011.04.014
- Bassis, J. N., & Jacobs, S. (2013). Diverse calving patterns linked to glacier geometry. Nature Geoscience, 6(10), 833–836. https://doi.org/10. 1038/ngeo1887
- Beaird, N., Straneo, F., & Jenkins, W. (2015). Spreading of Greenland meltwaters in the ocean revealed by noble gases. *Geophysical Research Letters*, 42, 7705–7713. https://doi.org/10.1002/(ISSN)1944-8007
- Beaird, N. L., Straneo, F., & Jenkins, W. (2018). Export of Strongly Diluted Greenland Meltwater From a Major Glacial Fjord. Geophysical Research Letters, 45(9), 4163–4170. https://doi.org/10.1029/2018GL077000
- Bendtsen, J., Mortensen, J., Lennert, K., & Rysgaard, S. (2015). Heat sources for glacial ice melt in a west Greenland tidewater outlet glacier fjord: The role of subglacial freshwater discharge. *Geophysical Research Letters*, 42(10), 4089–4095. https://doi.org/10.1002/ 2015GL063846
- Benn, D. I., Cowton, T. R., Todd, J., & Luckman, A. J. (2017). Glacier Calving in Greenland. Current Climate Change Reports, 3(4), 1–9. https://doi.org/10.1007/s40641-017-0070-1
- Benn, D. I., Hulton, N., & Mottram, R. H. (2007). 'Calving laws','sliding laws' and the stability of tidewater glaciers. *Annals of Glaciology*, 46, 123–130.



Bennartz, R., Shupe, M. D., Turner, D. D., Walden, V. P., Steffen, K., Cox, C. J., et al. (2013). July 2012 Greenland melt extent enhanced by low-level liquid clouds. *Nature*, 496(7443), 83–86. https://doi.org/10.1038/nature12002

Bevan, S. L., Luckman, A. J., & Murray, T. (2012). Glacier dynamics over the last quarter of a century at Helheim, Kangerdlugssuaq and 14 other major Greenland outlet glaciers. *The Cryosphere*, 6(5), 923–937. https://doi.org/10.5194/tc-6-923-2012

- Bevis, M. M., Harig, C., Khan, S. A., Brown, A. A., Simons, F. J., Willis, M. J., et al. (2019). Accelerating changes in ice mass within Greenland, and the ice sheet's sensitivity to atmospheric forcing. *Proceedings of the National Academy of Sciences*, 116(6), 1934–1939. https://doi. org/10.1073/pnas.1806562116
- Bjørk, A. A., Kjær, K. H., Korsgaard, N. J., Khan, S. A., Kjeldsen, K. K., Andresen, C. S., et al. (2012). An aerial view of 80 years of climate-related glacier fluctuations in southeast Greenland. *Nature Geoscience*, 5(6), 427–432. https://doi.org/10.1038/ngeo1481
- Bondzio, J. H., Morlighem, M., Seroussi, H., Kleiner, T., Rückamp, M., Mouginot, J., et al. (2017). The mechanisms behind Jakobshavn Isbrae's acceleration and mass loss: A 3-D thermomechanical model study. *Geophysical Research Letters*, 44(12), 6252–6260. https://doi. org/10.1002/2017GL073309
- Böning, C. W., Behrens, E., Biastoch, A., Getzlaff, K., & Bamber, J. L. (2016). Emerging impact of Greenland meltwater on deepwater formation in the North Atlantic Ocean. *Nature Geoscience*, 9(7), 523–527. https://doi.org/10.1038/ngeo2740
- Bougamont, M., Christoffersen, P., Hubbard, A. L., Fitzpatrick, A. A. W., Doyle, S. H., & Carter, S. P. (2014). Sensitive response of the Greenland Ice Sheet to surface melt drainage over a soft bed. *Nature Communications*, 5, 5052. https://doi.org/10.1038/ncomms6052
- Box, J. E. (2013). Greenland Ice Sheet Mass Balance Reconstruction. Part II: Surface Mass Balance (18402010)*. Journal of Climate, 26(18), 6974–6989. https://doi.org/10.1175/JCLI-D-12-00518.1
- Brinkerhoff, D., Truffer, M., & Aschwanden, A. (2017). Sediment transport drives tidewater glacier periodicity. *Nature Communications*, 8(1), 91. https://doi.org/10.1038/s41467-017-00095-5
- van den Broeke, M. R., Enderlin, E. M., Howat, I. M., Kuipers Munneke, P., Noël, B. P. Y., van De Berg, W. J., et al. (2016). On the recent contribution of the Greenland ice sheet to sea level change. *The Cryosphere*, *10*(5), 1933–1946. https://doi.org/10.5194/tc-10-1933-2016 van den Broeke, M. R., & Lenaerts, J. T. M. (2014). Contemporary (19602012) Evolution of the Climate and Surface Mass Balance of the
- Greenland Ice Sheet. Surveys in Geophysics, 35(5), 1155–1174. https://doi.org/10.1007/s10712-013-9261-z Brough, S., Carr, J. R., Ross, N., & Lea, J. M. (2019). Exceptional retreat of Kangerlussuaq Glacier, east Greenland, between 2016 and 2018. Frontiers in Earth Science, 7(123). https://doi.org/10.3389/feart.2019/00123
- Bunce, C., Carr, J. R., Nienow, P. W., Ross, N., & Killick, R. (2018). Ice front change of marine-terminating outlet glaciers in northwest and southeast Greenland during the 21st century. *Journal of Glaciology*, 64(246), 523–535. https://doi.org/10.1017/jog.2018.44
- Burton, J. C., Amundson, J. M., Abbot, D. S., Boghosian, A., Cathles, L. M., Correa-Legisos, S., et al. (2012). Laboratory investigations of iceberg capsize dynamics, energy dissipation and tsunamigenesis. *Journal of Geophysical Research*, 117, F01007. https://doi.org/10.1029/ 2011JF002055
- Cape, M. R., Straneo, F., Beaird, N., Bundy, R. M., & Charette, M. A. (2018). Nutrient release to oceans from buoyancy-driven upwelling at Greenland tidewater glaciers. *Nature Geoscience*, 12(1), 34–39. https://doi.org/10.1038/s41561-018-0268-4
- Carr, J. R., Stokes, C. R., & Vieli, A. (2013). Recent progress in understanding marine-terminating Arctic outlet glacier response to climatic and oceanic forcing: Twenty years of rapid change. *Progress in Physical Geography*, 37(4), 436–467. https://doi.org/10.1177/ 0309133313483163
- Carr, J. R., Vieli, A., & Stokes, C. R. (2013). Influence of sea ice decline, atmospheric warming, and glacier width on marine-terminating outlet glacier behavior in northwest Greenland at seasonal to interannual timescales. *Journal of Geophysical Research: Earth Surface*, 118, 1210–1226. https://doi.org/10.1002/jgrf.20088
- Carroll, D., Sutherland, D. A., Curry, B., Nash, J. D., Shroyer, E. L., Catania, G. A., et al. (2018). Subannual and Seasonal Variability of Atlantic-Origin Waters in Two Adjacent West Greenland Fjords. *Journal of Geophysical Research: Oceans*, 123, 6670–6687. https://doi. org/10.1029/2018JC014278
- Carroll, D., Sutherland, D. A., Hudson, B., Moon, T., Catania, G. A., Shroyer, E. L., et al. (2016). The impact of glacier geometry on meltwater plume structure and submarine melt in Greenland fjords. *Geophysical Research Letters*, 43, 9739–9748. https://doi.org/10.1002/ 2016GL070170
- Carroll, D., Sutherland, D. A., Shroyer, E. L., Nash, J. D., Catania, G. A., & Stearns, L. A. (2015). Modeling Turbulent Subglacial Meltwater Plumes: Implications for Fjord-Scale Buoyancy-Driven Circulation. *Journal of Physical Oceanography*, 45(8), 2169–2185. https://doi.org/10.1175/JPO-D-15-0033.1
- Carroll, D., Sutherland, D. A., Shroyer, E. L., Nash, J. D., Catania, G. A., & Stearns, L. A. (2017). Subglacial discharge-driven renewal of tidewater glacier fjords. *Journal of Geophysical Research: Oceans*, 122, 6611–6629. https://doi.org/10.1002/2017JC012962
- Cassotto, R., Fahnestock, M. A., Amundson, J. M., Truffer, M., & Joughin, I. R. (2015). Seasonal and interannual variations in ice melange and its impact on terminus stability, Jakobshavn Isbræ, Greenland. Journal of Glaciology, 61(225), 76–88. https://doi.org/10.3189/ 2015JoG13J235
- Catania, G. A., & Neumann, T. (2010). Persistent englacial drainage features in the Greenland Ice Sheet. *Geophysical Research Letters*, 37, L02501. https://doi.org/10.1029/2009GL041108
- Catania, G. A., & Paola, C. (2001). Braiding under glass. Geology, 29(3), 259-262.
- Catania, G. A., Stearns, L. A., Sutherland, D. A., Fried, M. J., Bartholomaus, T. C., Morlighem, M., et al. (2018). Geometric Controls on Tidewater Glacier Retreat in Central Western Greenland. *Journal of Geophysical Research: Earth Surface*, 123, 2024–2038. https://doi. org/10.1029/2017JF004499
- Chauché, N., Hubbard, A. L., Gascard, J. C., Box, J. E., Bates, R., Koppes, M. N., et al. (2014). Iceocean interaction and calving front morphology at two west Greenland tidewater outlet glaciers. *The Cryosphere*, 8(4), 1457–1468. https://doi.org/10.5194/tc-8-1457-2014

Chen, J. L., Wilson, C. R., & Tapley, B. D. (2011). Interannual variability of Greenland ice losses from satellite gravimetry. *Journal of Geophysical Research*, *116*, B07406. https://doi.org/10.1029/2010JB007789

Chen, X., Zhang, X., Church, J. A., Watson, C. S., King, M. A., Monselesan, D., et al. (2017). The increasing rate of global mean sea-level rise during 19932014. *Nature Climate Change*, 7, 492–495. https://doi.org/10.1038/nclimate3325

- Choi, Y., Morlighem, M., Rignot, E. J., Mouginot, J., & Wood, M. (2017). Modeling the Response of Nioghalvfjerdsfjorden and Zachariae Isstrøm Glaciers, Greenland, to Ocean Forcing Over the Next Century. *Geophysical Research Letters*, 44, 11,071–11,079. https://doi.org/ 10.1002/2017GL075174
- Choi, Y., Morlighem, M., Wood, M., & Bondzio, J. H. (2018). Comparison of four calving laws to model Greenland outlet glaciers. *The Cryosphere*, *12*(12), 3735–3746. https://doi.org/10.5194/tc-12-3735-2018

Christianson, K., Peters, L. E., Alley, R. B., Anandakrishnan, S., Jacobel, R. W., Riverman, K. L., et al. (2014). Dilatant till facilitates ice-stream flow in northeast Greenland. *Earth and Planetary Science Letters*, 401(C), 57–69. https://doi.org/10.1016/j.epsl.2014.05.060



Clark, P. U., Shakun, J. D., Marcott, S. A., Mix, A. C., Eby, M., Kulp, S., et al. (2016). Consequences of twenty-first century policy for multi-millennial climate and sea-level change. *Nature Climate Change*, 6(4), 360–369. https://doi.org/10.1038/nclimate2923

Clarke, T. S., & Echelmeyer, K. A. (1996). Seismic-reflection evidence for a deep subglacial trough beneath Jakobshavns Isbrae West Greenland. *Journal of Glaciology*, 43, 219–232.

Colgan, W., Rajaram, H., Abdalati, W., McCutchan, C., Mottram, R., Moosavi, M. S., & Grigsby, S. (2016). Glacier Crevasses: Observations Models and Mass Balance Implications. *Reviews of Geophysics*. https://doi.org/10.1002/(ISSN)1944-9208

Cook, A. J., Copeland, L., Noël, B. P. Y., Stokes, C. R., Bentley, M. J., Sharp, M. J., et al. (2019). Atmospheric forcing of rapid marine-terminating glacier retreat in the Canadian Arctic Archipelago. Science Advances, 5(3). https://doi.org/10.1126/sciadv.aau8507

Cook, A. J., Holland, P. R., Meredith, M. P., & Murray, T. (2016). Ocean forcing of glacier retreat in the western Antarctic Peninsula. *Science*, 353(6296), 283–286. https://doi.org/10.1126/science.aag0209

Cowton, T. R., Slater, D., Sole, A. J., Goldberg, D. N., & Nienow, P. W. (2015). Modeling the impact of glacial runoff on fjord circulation and submarine melt rate using a new subgrid-scale parameterization for glacial plumes. *Journal of Geophysical Research: Oceans, 120*, 796–812. https://doi.org/10.1002/2014JC010324

Cowton, T. R., Sole, A. J., Nienow, P. W., Slater, D., Wilton, D., & Hanna, E. (2016). Controls on the transport of oceanic heat to Kangerdlugssuaq Glacier, East Greenland. *Journal of Glaciology*, 62(236), 1167–1180. https://doi.org/10.1017/jog.2016.117

Csatho, B. M., Schenk, A. F., van der Veen, C. J., Babonis, G. S., Duncan, K., Rezvanbehbahani, S., et al. (2014). Laser altimetry reveals complex pattern of Greenland Ice Sheet dynamics. *Proceedings of the National Academy of Sciences*, 111(52), 18,478–18,483. https://doi.org/10.1073/pnas.1411680112

Das, S. B., Joughin, I. R., Behn, M. D., Howat, I. M., King, M. A., Lizarralde, D., & Bhatia, M. P. (2008). Fracture propagation to the base of the Greenland Ice Sheet during supraglacial lake drainage. *Science*, 320, 778–781. https://doi.org/10.1126/science.1153360

De Juan, J., Elósegui, P., Nettles, M., Larsen, T. B., Davis, J. L., Hamilton, G. S., et al. (2010). Sudden increase in tidal response linked to calving and acceleration at a large Greenland outlet glacier. *Geophysical Research Letters*, 37, L12501. https://doi.org/10.1029/2010GL043289

Delhasse, A., Fettweis, X., Kittel, C., Amory, C., & Agosta, C. (2018). Brief communication: Impact of the recent atmospheric circulation change in summer on the future surface mass balance of the Greenland Ice Sheet. *The Cryosphere*, 12(11), 3409–3418. https://doi.org/10.5194/tc-12-3409-2018

Dow, C. F., Hubbard, A. L., Booth, A. D., Doyle, S. H., Gusmeroli, A., & Kulessa, B. (2013). Seismic evidence of mechanically weak sediments underlying Russell Glacier, West Greenland. Annals of Glaciology, 54(64), 135–141. https://doi.org/10.3189/2013AoG64A032

Dowdeswell, J. A., & Vásquez, M. (2013). Submarine landforms in the fjords of southern Chile: implications for glacimarine processes and sedimentation in a mild glacier-influenced environment. *Quaternary Science Reviews*, 64(C), 1–19. https://doi.org/10.1016/j.quascirev. 2012.12.003

Ekstrom, G., Nettles, M., & Tsai, V. C. (2006). Seasonality and Increasing Frequency of Greenland Glacial Earthquakes. Science, 311(1756), 1756–1758. https://doi.org/10.1126/science.1122112

Enderlin, E. M., & Bartholomaus, T. C. (2019). Poor performance of a common crevasse model at marine-terminating glaciers. *The Cryosphere Discussions*, 1–19. https://doi.org/10.5194/tc-2019-128

Enderlin, E. M., Carrigan, C. J., Kochtitzky, W. H., Cuadros, A., Moon, T., & Hamilton, G. S. (2018). Greenland iceberg melt variability from high-resolution satellite observations. *The Cryosphere*, 12(2), 565–575. https://doi.org/10.5194/tc-12-565-2018

Enderlin, E. M., Hamilton, G. S., Straneo, F., & Sutherland, D. A. (2016). Iceberg meltwater fluxes dominate the freshwater budget in Greenland's iceberg-congested glacial fjords. *Geophysical Research Letters*, 43, 11,287–11,294. https://doi.org/10.1002/(ISSN)1944-8007

Enderlin, E. M., & Howat, I. M. (2013). Submarine melt rate estimates for floating termini of Greenland outlet glaciers (20002010). Journal of Glaciology, 59(213), 67–75. https://doi.org/10.3189/2013J0G12J049

Enderlin, E. M., Howat, I. M., Jeong, S., Noh, M.-J., van Angelen, J. H., & van den Broeke, M. R. (2014). An improved mass budget for the Greenland ice sheet. *Geophysical Research Letters*, 41, 866–872. https://doi.org/10.1002/2013GL059010

Enderlin, E. M., Howat, I. M., & Vieli, A. (2013). High sensitivity of tidewater outlet glacier dynamics to shape. The Cryosphere, 7(3), 1007–1015. https://doi.org/10.5194/tc-7-1007-2013

Enderlin, E. M., O'Neel, S., Bartholomaus, T. C., & Joughin, I. R. (2018). Evolving Environmental and Geometric Controls on Columbia Glacier's Continued Retreat. Journal of Geophysical Research: Earth Surface, 28, 2034–2018. https://doi.org/10.1029/2017JF004541

Enfield, D., Mestas-Nunez, A., & Trimble, P. (2001). The Atlantic Multidecadal Oscillation and its relationship to rainfall and river flows in the continental US. *Geophysical Research Letters*, 28(10), 2077–2080.

Everett, A., Kohler, J., Sundfjord, A., Kovacs, K. M., Torsvik, T., Pramanik, A., et al. (2018). Subglacial discharge plume behaviour revealed by CTD-instrumented ringed seals. *Scientific Reports*, 8(1), 1–10. https://doi.org/10.1038/s41598-018-31875-8

Ezhova, E., Cenedese, C., & Brandt, L. (2018). Dynamics of Three-Dimensional Turbulent Wall Plumes and Implications for Estimates of Submarine Glacier Melting. Journal of Physical Oceanography, 48(9), 1941–1950. https://doi.org/10.1175/JPO-D-17-0194.1

Fahnestock, M. A., Scambos, T. A., Moon, T., Gardner, A., Haran, T., & Klinger, M. (2016). Rapid large-area mapping of ice flow using Landsat 8. *Remote Sensing of Environment*, 185, 84–94. https://doi.org/10.1016/j.rse.2015.11.023

Farinotti, D., Huss, M., Fürst, J. J., Landmann, J., Machguth, H., Maussion, F., & Pandit, A. (2019). A consensus estimate for the ice thickness distribution of all glaciers on Earth. Nature Geoscience, 12(3), 168–173. https://doi.org/10.1038/s41561-019-0300-3

Felikson, D., Bartholomaus, T. C., Catania, G. A., Korsgaard, N. J., Kjær, K. H., Morlighem, M., et al. (2017). Inland thinning on the Greenland ice sheet controlled by outlet glacier geometry. *Nature Geoscience*, 10(5), 366–369. https://doi.org/10.1038/ngeo2934

Fenty, I., Willis, J., Khazendar, A., Dinardo, S., Forsberg, R. R., Fukumori, I., et al. (2016). Oceans Melting Greenland: Early Results from NASA's Ocean-Ice Mission in Greenland. *Oceanography*, 29(4), 72–83. https://doi.org/10.5670/oceanog.2016.100

Fettweis, X., Box, J. E., Agosta, C., Amory, C., Kittel, C., Lang, C., et al. (2017). Reconstructions of the 19002015 Greenland ice sheet surface mass balance using the regional climate MAR model. *The Cryosphere*, *11*(2), 1015–1033. https://doi.org/10.5194/tc-11-1015-2017

Forster, R. R., Box, J. E., van den Broeke, M. R., Miège, C., Burgess, E. W., van Angelen, J. H., et al. (2014). Extensive liquid meltwater storage in firm within the Greenland ice sheet. *Nature Geoscience*, 7(2), 95–98. https://doi.org/10.1038/ngeo2043

Fraser, N. J., & Inall, M. E. (2018). Influence of Barrier Wind Forcing on Heat Delivery Toward the Greenland Ice Sheet. Journal of Geophysical Research: Oceans, 5, 300–326. https://doi.org/10.1002/2017JC013464

Frederick, E., Manizade, S., Martin, C., Sonntag, J., Swift, R. N., Thomas, R. H., et al. (2000). Greenland Ice Sheet: High-Elevation Balance and Peripheral Thinning. Science, 289(5478), 428–430.

Fried, M. J., Carroll, D., Catania, G. A., Sutherland, D. A., Stearns, L. A., Shroyer, E. L., & Nash, J. D. (2019). Distinct frontal ablation processes drive heterogeneous submarine terminus morphology. *Geophysical Research Letters*, 46, 12,083–12,091. https://doi.org/10. 1029/2019GL083980 Fried, M. J., Catania, G. A., Bartholomaus, T. C., Duncan, D., Davis, M., Stearns, L. A., et al. (2015). Distributed subglacial discharge drives significant submarine melt at a Greenland tidewater glacier. *Geophysical Research Letters*, 42, 9328–9336. https://doi.org/10.1002/ 2015GL065806

Fried, M. J., Catania, G. A., Stearns, L. A., Sutherland, D. A., Bartholomaus, T. C., Shroyer, E. L., & Nash, J. D. (2018). Reconciling Drivers of Seasonal Terminus Advance and Retreat at 13 Central West Greenland Tidewater Glaciers. *Journal of Geophysical Research: Earth* Surface, 123, 1590–1607. https://doi.org/10.1029/2018JF004628

- Fürst, J. J., Goelzer, H., & Huybrechts, P. (2015). Ice-dynamic projections of the Greenland ice sheet in response to atmospheric and oceanic warming. *The Cryosphere*, 9(3), 1039–1062. https://doi.org/10.5194/tc-9-1039-2015
- Fyke, J., Sergienko, O., Löfverström, M., Price, S., & Lenaerts, J. T. M. (2018). An Overview of Interactions and Feedbacks Between Ice Sheets and the Earth System. *Reviews of Geophysics*, 56, 361–408. https://doi.org/10.1029/2018RG000600
- Fyke, J. G., Vizcaíno, M., Lipscomb, W., & Price, S. F. (2014). Future climate warming increases Greenland ice sheet surface mass balance variability. *Geophysical Research Letters*, 41, 470–475. https://doi.org/10.1002/(ISSN)1944-8007
- Gabel-Jorgensen, C. (1935). Dr. Knud Rasmussen's Contribution to the Exploration of the South-East Coast of Greenland. *The Geographical Journal*, 86(1), 32–49.
- GEBCO Compilation Group (2019). GEBCO 2019, https://doi.org/10.5285/836f016a-33be-6ddc-e053-6c86abc0788e.
- Gillard, L. C., Hu, X., Myers, P. G., & Bamber, J. L. (2016). Meltwater pathways from marine terminating glaciers of the Greenland ice sheet. Geophysical Research Letters, 43, 10,873–10,882. https://doi.org/10.1002/(ISSN)1944-8007
- Gladish, C. V., Holland, D. M., & Lee, C. M. (2015). Oceanic Boundary Conditions for Jakobshavn Glacier. Part II: Provenance and Sources of Variability of Disko Bay and Ilulissat Icefjord Waters, 19902011. *Journal of Physical Oceanography*, 45(1), 33–63. https://doi.org/10.1175/ JPO-D-14-0045.1
- Gladish, C. V., Holland, D. M., Rosing-Asvid, A., Behrens, J. W., & Boje, J. (2015). Oceanic boundary conditions for Jakobshavn Glacier. Part I: Variability and renewal of Ilulissat Icefjord waters, 200114. *Journal of Physical Oceanography*, 45. https://doi.org/10.1175/ JPO-D-14-0044.s1
- Glasser, N. F., & Scambos, T. A. (2008). A structural glaciological analysis of the 2002 Larsen B ice-shelf collapse. *Journal of Glaciology*, 54(184), 3–17.
- Gogineni, S. P., Yan, J. B., Paden, J. D., Leuschen, C. J., Li, J., Rodriquez-Morales, F., et al. (2014). Bed topography of Jakobshavn Isbræ, Greenland, and Byrd Glacier, Antarctica. *Journal of Glaciology*, 60(223), 813–833. https://doi.org/10.3189/2014JoG14J129
- Goldberg, D. N., Heimbach, P., Joughin, I. R., & Smith, B. E. (2015). Committed retreat of Smith, Pope, and Kohler Glaciers over the next 30 years inferred by transient model calibration. *The Cryosphere*, 9(6), 2429–2446. https://doi.org/10.5194/tc-9-2429-2015
- Golledge, N. R., Keller, E. D., Gomez, N., Naughten, K. A., Bernales, J., Trusel, L. D., & Edwards, T. L. (2019). Global environmental consequences of twenty-first-century ice-sheet melt. *Nature*, 566(7742), 1–23. https://doi.org/10.1038/s41586-019-0889-9
- de la Guardia, L. C., Hu, X., & Myers, P. G. (2015). Potential positive feedback between Greenland Ice Sheet melt and Baffin Bay heat content on the west Greenland shelf. *Geophysical Research Letters*, 42, 4922–4930. https://doi.org/10.1002/(ISSN)1944-8007
- Hanna, E., Cropper, T. E., Hall, R. J., & Cappelen, J. (2016). Greenland Blocking Index 1851-2015: a regional climate change signal. International Journal of Climatology, 36(15), 4847–4861. https://doi.org/10.1002/joc.4673
- Hanna, E., Fettweis, X., Mernild, S. H., Cappelen, J., Ribergaard, M. H., Shuman, C. A., et al. (2013). Atmospheric and oceanic climate forcing of the exceptional Greenland ice sheet surface melt in summer 2012. *International Journal of Climatology*, 34(4), 1022–1037. https://doi.org/10.1002/joc.3743
- Hanna, E., Jones, J. M., Cappelen, J., Mernild, S. H., Wood, L., Steffen, K., & Huybrechts, P. (2012). The influence of North Atlantic atmospheric and oceanic forcing effects on 1900-2010 Greenland summer climate and ice melt/runoff. *International Journal of Climatology*, 33(4), 862–880. https://doi.org/10.1002/joc.3475
- Harden, B. E., Renfrew, I. A., & Petersen, G. N. (2011). A Climatology of Wintertime Barrier Winds off Southeast Greenland. Journal of Climate, 24(17), 4701–4717. https://doi.org/10.1175/2011JCLI4113.1
- Harper, J. T., Humphrey, N. F., Meierbachtol, T. W., Graly, J. A., & Fischer, U. H. (2017). Borehole measurements indicate hard bed conditions, Kangerlussuaq sector, western Greenland Ice Sheet. Journal of Geophysical Research: Earth Surface, 122, 1605–1618. https:// doi.org/10.1002/2017JF004201
- Harrington, J. A., Humphrey, N. F., & Harper, J. T. (2015). Temperature distribution and thermal anomalies along a flowline of the Greenland ice sheet. *Annals of Glaciology*, 56(70), 98–104. https://doi.org/10.3189/2015AoG70A945
- Haseloff, M., & Sergienko, O. (2018). The effect of buttressing on grounding line dynamics. Journal of Glaciology, 64(245), 417-431.
- Heimbach, P., Fukumori, I., Hill, C. N., Ponte, R. M., Stammer, D., Wunsch, C., et al. (2019). Putting It All Together: Adding Value to the Global Ocean and Climate Observing Systems With Complete Self-Consistent Ocean State and Parameter Estimates. Frontiers in Marine Science, 6, 769–710. https://doi.org/10.3389/fmars.2019.00055
- Hofer, S., Tedstone, A. J., Fettweis, X., & Bamber, J. L. (2017). Decreasing cloud cover drives the recent mass loss on the Greenland Ice Sheet. Science Advances, 3(6), e1700584. https://doi.org/10.1126/sciadv.1700584
- Hoffman, M. J., Andrews, L. C., Price, S. A., Catania, G. A., Neumann, T., Lüthi, M. P., et al. (2016). Greenland subglacial drainage evolution regulated by weakly connected regions of the bed. *Nature Communications*, 7(13903). https://doi.org/10.1038/ncomms13903
- Hofstede, C., Christoffersen, P., Hubbard, B., Doyle, S. H., Young, T. J., Diez, A., et al. (2018). Physical Conditions of Fast Glacier Flow: 2 Variable Extent of Anisotropic Ice and Soft Basal Sediment From Seismic Reflection Data Acquired on Store Glacier, West Greenland. Journal of Geophysical Research: Earth Surface, 123, 349–362. https://doi.org/10.1002/2017JF004297
- Holland, D. M., & Jenkins, A. (1999). Modeling Thermodynamic Ice-Ocean Interactions at the Base of an Ice Shelf. Journal of Physical Oceanography, 29(8), 1787–1800. https://doi.org/10.1175/1520-0485(1999)029<1787:MTIOIA>2.0.CO;2
- Holland, D. M., Thomas, R. H., De Young, B., Ribergaard, M. H., & Lyberth, B. (2008). Acceleration of Jakobshavn Isbræ triggered by warm subsurface ocean waters. *Nature Geoscience*, 1, 659–664. https://doi.org/10.1038/ngeo316
- Hopwood, M. J., Carroll, D., Browning, T. J., Meire, L., Mortensen, J., Krisch, S., & Achterberg, E. P. (2018). Non-linear response of summertime marine productivity to increased meltwater discharge around Greenland. *Nature Communications*, 9(3256). https:// doi.org/10.1038/s41467-018-05488-8
- Howat, I. M. (2017). MEaSURES Greenland Ice Velocity: Selected Glacier Site Velocity Maps from Optical Images, Version 2., NASA National Snow and Ice Data Center Distributed Active Archive Center, Boulder, Colorado, USA, https://doi.org/10.5067/ VM5DZ20MYF5C, doi: https://doi.org/10.5067/VM5DZ20MYF5C.
- Howat, I. M., Ahn, Y., Joughin, I. R., van den Broeke, M. R., Lenaerts, J. T. M., & Smith, B. E. (2011). Mass balance of Greenland's three largest outlet glaciers, 20002010. *Geophysical Research Letters*, 38, L12501. https://doi.org/10.1029/2011GL047565
- Howat, I. M., & Eddy, A. (2011). Multi-decadal retreat of Greenland's marine-terminating glaciers. Journal of Glaciology, 57(203), 1-8.

- Howat, I. M., Joughin, I. R., Fahnestock, M. A., Smith, B. E., & Scambos, T. A. (2008). Synchronous retreat and acceleration of southeast Greenland outlet glaciers 200006: Ice dynamics and coupling to climate. *Journal of Glaciology*, 54(187), 646–660.
- Howat, I. M., Joughin, I. R., & Scambos, T. A. (2007). Rapid Changes in Ice Discharge from Greenland Outlet Glaciers. *Science*, 315(5818), 1559–1561.
- Howat, I. M., Negrete, A., & Smith, B. E. (2014). The Greenland Ice Mapping Project (GIMP) land classification and surface elevation data sets. *The Cryosphere*, 8(4), 1509–1518. https://doi.org/10.5194/tc-8-1509-2014
- Howat, I. M., Smith, B. E., & Scambos, T. A. (2008). Rates of southeast Greenland ice volume loss from combined ICESat and ASTER observations. *Geophysical Research Letters*, 35, L17505. https://doi.org/10.1029/2008GL034496
- Hughes, T. J. (1975). The West Antartic Ice Sheet: Instability, disintegration and initiation of ice ages. *Reviews of Geophysics and Space Physics*, 13(4), 502–526.
- Iken, A., Echelmeyer, M., Harrison, W., & Funk, M. (2017). Mechanisms of fast flow in Jakobshavns Isbræ, West Greenland: Part I Measurements of temperature and water level in deep boreholes. *Journal of Glaciology*, 39(131), 15–25. https://doi.org/10.3189/ S0022143000015689
- Inall, M. E., Murray, T., Cottier, F. R., Scharrer, K., Boyd, T. J., Heywood, K. J., & Bevan, S. L. (2014). Oceanic heat delivery via Kangerdlugssuaq Fjord to the south-east Greenland ice sheet. *Journal of Geophysical Research: Oceans*, 119, 631–645. https://doi.org/10.1002/ 2013JC009295
- Jackson, R. H., Lentz, S. J., & Straneo, F. (2018). The Dynamics of Shelf Forcing in Greenlandic Fjords. *Journal of Physical Oceanography*, 48(11), 2799–2827. https://doi.org/10.1175/JPO-D-18-0057.1
- Jackson, R. H., Nash, J. D., Kienholz, C., Sutherland, D., Amundson, J. M., Motyka, R. J., et al. (2019). Meltwater intrustions revela mechanisms for rapid submarine melt at a tidewater glacier. *Geophysical Research Letters*.
- Jackson, R. H., Shroyer, E. L., Nash, J. D., Sutherland, D. A., Carroll, D., Fried, M. J., et al. (2017). Near-glacier surveying of a subglacial discharge plume: Implications for plume parameterizations. *Geophysical Research Letters*, 44, 6886–6894. https://doi.org/10.1002/ 2017GL073602
- Jackson, R. H., & Straneo, F. (2016). Heat, Salt, and Freshwater Budgets for a Glacial Fjord in Greenland. *Journal of Physical Oceanography*, 46(9), 2735–2768. https://doi.org/10.1175/JPO-D-15-0134.1
- Jackson, R. H., Straneo, F., & Sutherland, D. A. (2014). Externally forced fluctuations in ocean temperature at Greenland glaciers in non-summer months. *Nature Geoscience*, 7(7), 503–508. https://doi.org/10.1038/ngeo2186
- Jaeger, J. M., & Koppes, M. N. (2016). The role of the cryosphere in source-to-sink systems. *Earth Science Reviews*, 153(C), 43–76. https://doi.org/10.1016/j.earscirev.2015.09.011
- James, T. D., Murray, T., Selmes, N., Scharrer, K., & O'Leary, M. (2014). Buoyant flexure and basal crevassing in dynamic mass loss at Helheim Glacier. Nature Geoscience, 7(8), 593–596. https://doi.org/10.1038/ngeo2204
- Jenkins, A. (2011). Convection-driven melting near the grounding lines of ice shelves and tidewater glaciers. Journal of Physical Oceanography, 41(12), 2279–2294. https://doi.org/10.1175/JPO-D-11-03.1
- Jenkins, A., Dutrieux, P., Jacobs, S., McPhail, S. D., Perrett, J. R., Webb, A. T., & White, D. (2010). Observations beneath Pine Island Glacier in West Antarctica and implications for its retreat. *Nature Geoscience*, 3(7), 468–472. https://doi.org/10.1038/ngeo890
- Joughin, I. R., Das, S. B., King, M. A., Smith, B. E., Howat, I. M., & Moon, T. (2008). Seasonal speedup along the western flank of the Greenland Ice Sheet. Science, 320(5877), 781–783. https://doi.org/10.1126/science.1153288
- Joughin, I. R., Howat, I. M., Fahnestock, M. A., Smith, B. E., Krabill, W. B., Alley, R. B., et al. (2008). Continued evolution of Jakobshavn Isbrae following its rapid speedup. Journal of Geophysical Research, 113, F04006. https://doi.org/10.1029/2008JF001023
- Joughin, I. R., Smith, B. E., Howat, I. M., Scambos, T. A., & Moon, T. (2010). Greenland flow variability from ice-sheet-wide velocity mapping. Journal of Glaciology, 56(197), 415–430.
- Joughin, I. R., Smith, B. E., & Schoof, C. G. (2019). Regularized Coulomb Friction Laws for Ice Sheet Sliding: Application to Pine Island Glacier, Antarctica. Geophysical Research Letters, 310, 456–458. https://doi.org/10.1029/2019GL082526
- Kanna, N., Sugiyama, S., Ohashi, Y., Sakakibara, D., Fukamachi, Y., & Nomura, D. (2018). Upwelling of Macronutrients and Dissolved Inorganic Carbon by a Subglacial Freshwater Driven Plume in Bowdoin Fjord, Northwestern Greenland. *Journal of Geophysical Research: Biogeosciences*, 123, 1666–1682. https://doi.org/10.1029/2017JG004248

Kessler, M. A., Anderson, R. S., & Briner, J. P. (2008). Fjord insertion into continental margins driven by topographic steering of ice. Nature Geoscience, 1(6), 365–369. https://doi.org/10.1038/ngeo201

- Khan, S. A., Wahr, J. J., Bevis, M. M., Velicogna, I., & Kendrick, E. E. (2010). Spread of ice mass loss into northwest Greenland observed by GRACE and GPS. *Geophysical Research Letters*, 37, L06501. https://doi.org/10.1029/2010GL042460
- Khazendar, A., Fenty, I. G., Carroll, D., Gardner, A., Lee, C. M., Fukumori, I., et al. (2019). Interruption of two decades of Jakobshavn Isbrae acceleration and thinning as regional ocean cools. *Nature Geoscience*, *12*. https://doi.org/10.1038/s41561-019-0329-3
- Khazendar, A., Rignot, E. J., & Larour, E. Y. (2007). Larsen B Ice Shelf rheology preceding its disintegration inferred by a control method. *Geophysical Research Letters*, 34, L19503. https://doi.org/10.1029/2007GL030980
- Kimura, S., Holland, P. R., Jenkins, A., & Piggott, M. D. (2014). The Effect of Meltwater Plumes on the Melting of a Vertical Glacier Face. Journal of Physical Oceanography, 44(12), 3099–3117. https://doi.org/10.1175/JPO-D-13-0219.1
- King, M. D., Howat, I. M., Jeong, S., Noh, M. J., Wouters, B., Noël, B. P. Y., & van den Broeke, M. R. (2018). Seasonal to decadal variability in ice discharge from the Greenland Ice Sheet. *The Cryosphere*, 12(12), 3813–3825. https://doi.org/10.5194/tc-12-3813-2018
- Krabill, W. B., Hanna, E., Huybrechts, P., Abdalati, W., Cappelen, J., Csatho, B. M., et al. (2004). Greenland Ice Sheet: increased coastal thinning. *Geophysical Research Letters*, *31*, L24402. https://doi.org/10.1029/2004GL021533

Larour, E. Y., Ivins, E. R., & Adhikari, S. (2017). Should coastal planners have concern over where land ice is melting? *Science Advances*, 3(11). https://doi.org/10.1126/sciadv.1700537

- Lea, J. M. (2018). Google Earth Engine Digitisation Tool (GEEDiT), and Margin change Quantification Tool (MaQiT) & amp;ndash; simple tools for the rapid mapping and quantification of changing Earth surface margins. *Earth Surface Dynamics*, 6, 551–561. https://doi.org/10.5194/esurf-2018-24
- Lemke, P., Ren, J., Alley, R. B., Allison, I., Carrasco, J., Flato, G., et al. (2007). Observations: Changes in Snow, Ice and Frozen Ground. In S. Solomon, D. Qin, M. Manning, Z. Chen, M. Marquis, K. B. Averyt, M. Tignor, & H. L. Miller (Eds.), Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 1–48). Cambridge, United Kingdom: Cambridge Univ Press.
- Levitus, S., & Antonov, J. (2005). Warming of the world ocean, 19552003. Geophysical Research Letters, 32, L02604. https://doi.org/10.1029/2004GL021592
- Lewis, S. M., & Smith, L. C. (2009). Hydrologic drainage of the Greenland Ice Sheet. *Hydrological Processes*, 23(14), 2004–2011. https://doi. org/10.1002/hyp.7343

Lloyd, J. M., Moros, M., Perner, K., Telford, R. J., Kuijpers, A., Jansen, E., & McCarthy, D. J. (2011). A 100 yr record of ocean temperature control on the stability of Jakobshavn Isbrae, West Greenland. *Geology*, 39(9), 867–870. https://doi.org/10.1130/G32076.1

Logan, L., Catania, G. A., Lavier, L., & Choi, E. (2013). A novel method for predicting fracture in floating ice. *Journal of Glaciology*, 59(216), 750–758. https://doi.org/10.3189/2013JoG12J210

Luckman, A. J., Benn, D. I., Cottier, F. R., Bevan, S. L., Nilsen, F., & Inall, M. E. (2015). Calving rates at tidewater glaciers vary strongly with ocean temperature. *Nature Communications*, 6, 8566. https://doi.org/10.1038/ncomms9566

Luo, H., Castelao, R. M., Rennermalm, A. K., Tedesco, M., Bracco, A., Yager, P. L., & Mote, T. L. (2016). Oceanic transport of surface meltwater from the southern Greenland ice sheet. *Nature Geoscience*, 9(7), 528–532. https://doi.org/10.1038/ngeo2708

Lüthi, M. P., Funk, M., Iken, A., Gogineni, S. P., & Truffer, M. (2002). Mechanisms of Fast Flow in Jakobshavns Isbrae, Greenland; Part III: Measurements of ice deformation, temperature and cross-borehole conductivity in boreholes to the bedrock. *Journal of Glaciology*, *48*(162), 369–385.

Lüthi, M. P., Ryser, C., Andrews, L. C., Catania, G. A., Funk, M., Hawley, R. L., et al. (2015). Heat sources within the Greenland Ice Sheet: dissipation, temperate paleo-firn and cryo-hydrologic warming. *The Cryosphere*, 9(1), 245–253. https://doi.org/10.5194/tc-9-245-2015 MacAyeal, D. R. (1985). Evolution of tidally triggered meltwater plumes below ice shelves. *Antarctic Research Series*, 43, 133–143.

MacGregor, J. A., Fahnestock, M. A., Catania, G. A., Aschwanden, A., Clow, G. D., Colgan, W., et al. (2016). A synthesis of the basal thermal

state of the Greenland Ice Sheet. Journal of Geophysical Research: Earth Surface, 121, 1328–1350. https://doi.org/10.1002/2015JF003803 Magorrian, S. J., & Wells, A. J. (2016). Turbulent plumes from a glacier terminus melting in a stratified ocean. Journal of Geophysical Research: Oceans, 121, 4670–4696. https://doi.org/10.1002/2015JC011160

Mankoff, K. D., Colgan, W., Solgaard, A., Karlsson, N. B., Ahlstrøm, A., van As, D., et al. (2019). Greenland Ice Sheet solid ice discharge from 1986 through 2017. Earth System Science Data, 11, 769–786. https://doi.org/10.5194/essd-11-769-2019

Mankoff, K. D., Straneo, F., Cenedese, C., Das, S. B., Richards, C. G., & Singh, H. (2016). Structure and dynamics of a subglacial discharge plume in a Greenlandic fjord. *Journal of Geophysical Research: Oceans*, 121, 8670–8688. https://doi.org/10.1002/2016JC011764

Marzeion, B., Cogley, J. G., Richter, K., & Parkes, D. (2014). Attribution of global glacier mass loss to anthropogenic and natural causes. Science, 345(6199), 919–921. https://doi.org/10.1126/science.1254702

Mattingly, K. S., Mote, T. L., & Fettweis, X. (2018). Atmospheric River Impacts on Greenland Ice Sheet Surface Mass Balance. Journal of Geophysical Research: Atmospheres, 123, 8538–8560. https://doi.org/10.1029/2018JD028714

McFadden, E. M., Howat, I. M., Joughin, I. R., Smith, B. E., & Ahn, Y. (2011). Changes in the dynamics of marine terminating outlet glaciers in west Greenland (20002009). Journal of Geophysical Research, 116, F01005. https://doi.org/10.1029/2010JF001757

McNabb, R. W., Hock, R., & Huss, M. (2015). Variations in Alaska tidewater glacier frontal ablation, 1985-2013. Journal of Geophysical Research: Earth Surface, 120, 120–136. https://doi.org/10.1002/2014JF003276

McPhee, M. G., Maykut, G. A., & Morison, J. (1987). Dynamics and thermodynamics of the ice/upper ocean system in the marginal ice zone of the Greenland Sea. *Journal of Geophysical Research*, 92(C7), 7017–7031.

Meierbachtol, T. W., Harper, J. T., & Humphrey, N. F. (2013). Basal Drainage System Response to Increasing Surface Melt on the Greenland Ice Sheet. Science, 341(6147), 777–779. https://doi.org/10.1126/science.1235905

Meire, L., Mortensen, J., Meire, P., Juul-Pedersen, T., Sejr, M. K., Rysgaard, S., et al. (2017). Marine-terminating glaciers sustain high productivity in Greenland fjords. *Global Change Biology*, 23(12), 5344–5357. https://doi.org/10.1111/gcb.13801

Meire, L., Søgaard, D. H., Mortensen, J., Meysman, F. J. R., Soetaert, K., Arendt, K. E., et al. (2015). Glacial meltwater and primary production are drivers of strong CO2 uptake in fjord and coastal waters adjacent to the Greenland Ice Sheet. *Biogeosciences*, 12(8), 2347–2363. https://doi.org/10.5194/bg-12-2347-2015

Mengel, M., Levermann, A., Frieler, K., Robinson, A., Marzeion, B., & Winkelmann, R. (2016). Future sea level rise constrained by observations and long-term commitment. *Proceedings of the National Academy of Sciences*, 113(10), 2597–2602. https://doi.org/10.1073/ pnas.1500515113

Mercer, J. H. (1978). West Antarctic Ice Sheet and CO2 greenhouse effect: a threat of disaster. Nature, 271, 321-325.

Mernild, S. H., Mote, T. L., & Liston, G. E. (2011). Greenland ice sheet surface melt extent and trends: 1960-2010. Journal of Glaciology, 57(204), 621–628.

Millan, R., Rignot, E. J., Mouginot, J., Wood, M., Bjørk, A. A., & Morlighem, M. (2018). Vulnerability of Southeast Greenland Glaciers to Warm Atlantic Water From Operation IceBridge and Ocean Melting Greenland Data. *Geophysical Research Letters*, 45, 2688–2696. https://doi.org/10.1002/2017GL076561

Minchew, B. M., Meyer, C. R., Pegler, S. S., Lipovsky, B. P., Rempel, A. W., Gudmundsson, G. H., & Iverson, N. R. (2019). Comment on "Friction at the bed does not control fast glacier flow". *Science*, 363(6427), eaau6055. https://doi.org/10.1126/science.aau6055

Mitrovica, J. X., Gomez, N., & Clark, P. U. (2009). The sea-level fingerprint of West Antarctic collapse. *Science*, 323(5915), 753.

Mitrovica, J. X., Hay, C. C., Kopp, R. E., Harig, C., & Latychev, K. (2018). Quantifying the Sensitivity of Sea Level Change in Coastal Localities to the Geometry of Polar Ice Mass Flux. *Journal of Climate*, *31*(9), 3701–3709. https://doi.org/10.1175/JCLI-D-17-0465.1

Mohajerani, Y., M. Wood, I. Velicogna, and E. J. Rignot (2018), Detection of Glacier Calving Margins with Convolutional Neural Networks: A Case Study, Creative Commons, pp. 1–12, doi: 10.20944/preprints201811.0529.v1.

Moon, T., & Joughin, I. R. (2008). Changes in ice front position on Greenland's outlet glaciers from 1992 to 2007. Journal of Geophysical Research, 113, F02022. https://doi.org/10.1029/2007JF000927

Moon, T., Joughin, I. R., & Smith, B. E. (2015). Seasonal to multiyear variability of glacier surface velocity, terminus position, and sea ice/ice mélange in northwest Greenland. *Journal of Geophysical Research: Earth Surface*, 120, 818–833. https://doi.org/10.1002/2015JF003494

Moon, T., Joughin, I. R., Smith, B. E., & Howat, I. M. (2012). 21st-Century Evolution of Greenland Outlet Glacier Velocities. Science, 336(6081), 576–578. https://doi.org/10.1126/science.1219985

Moon, T., Joughin, I. R., Smith, B. E., van den Broeke, M. R., van De Berg, W. J., Noël, B. P. Y., & Usher, M. (2014). Distinct patterns of seasonal Greenland glacier velocity. *Geophysical Research Letters*, 41, 7209–7216. https://doi.org/10.1002/2014GL061836

Moon, T., Sutherland, D. A., Carroll, D., Felikson, D., Kehrl, L., & Straneo, F. (2017). Subsurface iceberg melt key to Greenland fjord freshwater budget. *Nature Geoscience*, 11(1), 49–54. https://doi.org/10.1038/s41561-017-0018-z

Morlighem, M., Bondzio, J. H., Seroussi, H., Rignot, E. J., Larour, E. Y., Humbert, A., & Rebuffi, S. (2016). Modeling of Store Gletscher's calving dynamics West Greenland, in response to ocean thermal forcing. *Geophysical Research Letters*, 43, 2659–2666. https://doi.org/ 10.1002/2016GL067695

Morlighem, M., Rignot, E. J., & Willis, J. (2016). Improving Bed Topography Mapping of Greenland Glaciers Using NASA's Oceans Melting Greenland (OMG) Data. *Oceanography*, 29(4), 62–71. https://doi.org/10.5670/oceanog.2016.99

Morlighem, M., Williams, C. N., Rignot, E. J., An, L., Arndt, J. E., Bamber, J. L., et al. (2017). BedMachine v3: Complete Bed Topography and Ocean Bathymetry Mapping of Greenland From Multibeam Echo Sounding Combined With Mass Conservation. *Geophysical Research Letters*, 44, 11,051–11,061. https://doi.org/10.1002/2017GL074954

- Mortensen, J., Bendtsen, J., Lennert, K., & Rysgaard, S. (2014). Seasonal variability of the circulation system in a west Greenland tidewater outlet glacier fjord, Godthåbsfjord (64°N). Journal of Geophysical Research: Earth Surface, 119, 2591–2603. https://doi.org/10. 1002/2014JF003267
- Morton, B. R., Taylor, G. I., & Turner, J. S. (1956). Turbulent gravitational convection from maintained and instantaneous sources, Proceedings of the Royal Society A: Mathematical. *Physical and Engineering Sciences*, 234(1196), 1–24.
- Motyka, R. J., Dryer, W. P., Amundson, J. M., Truffer, M., & Fahnestock, M. A. (2013). Rapid submarine melting driven by subglacial discharge, LeConte Glacier, Alaska. *Geophysical Research Letters*, 40, 5153–5158. https://doi.org/10.1002/grl.51011
- Motyka, R. J., Hunter, L. E., Echelmeyer, K. A., & Connor, C. (2003). Submarine melting at the terminus of a temperate tidewater glacier, LeConte Glacier, Alaska, U.S.A. Annals of Glaciology, 36, 57–65.
- Motyka, R. J., Truffer, M., Fahnstock, M., Mortensen, J., Rysgaard, S., & Howat, I. M. (2011). Submarine melting of the 1985 Jakobshavn Isbræ floating tongue and the triggering of the current retreat. *Wiley Online Library*, *116*, F01007. https://doi.org/10.1029/2009JF001632
- Motyka, R. J., Truffer, M., Kuriger, E. M., & Bucki, A. K. (2006). Rapid erosion of soft sediments by tidewater glacier advance: Taku Glacier, Alaska, USA. Geophysical Research Letters, 33, L24504. https://doi.org/10.1029/2006GL028467
- Mouginot, J., Rignot, E. J., Bjørk, A. A., van den Broeke, M. R., Millan, R., Morlighem, M., et al. (2019). Forty-six years of Greenland Ice Sheet mass balance from 1972 to 2018. Proceedings of the National Academy of Sciences, 116(19), 9239–9244. https://doi.org/10.7280/ D1MM37
- Murray, T., Scharrer, K., James, T. D., Dye, S. R., Hanna, E., Booth, A. D., et al. (2010). Ocean regulation hypothesis for glacier dynamics in southeast Greenland and implications for ice sheet mass changes. *Journal of Geophysical Research*, *115*, F03026. https://doi.org/10. 1029/2009JF001522
- Murray, T., Scharrer, K., Selmes, N., Booth, A. D., James, T. D., Bevan, S. L., et al. (2015). Extensive Retreat of Greenland Tidewater Glaciers, 20002010. Arctic, Antarctic, and Alpine Research, 47(3), 427–447. https://doi.org/10.1657/AAAR0014-049
- Murray, T., Selmes, N., James, T. D., Edwards, S., Martin, I., O'Farrell, T., et al. (2015). Dynamics of glacier calving at the ungrounded margin of Helheim Glacier, southeast Greenland. *Journal of Geophysical Research*, *120*, 964–982. https://doi.org/10.1002/2015JF003531 Nghiem, S. V., Hall, D. K., Mote, T. L., Tedesco, M., Albert, M. R., Keegan, K., et al. (2012). The extreme melt across the Greenland ice sheet
- In Structures and Structures and
- NICK, F. M., van der veen, C. J., & Oerlemans, J. (2007). Controls on advance of tidewater glaciers: Results from numerical modeling applied to Columbia Glacier. *Journal of Geophysical Research*, 112, F03S24. https://doi.org/10.1029/2006JF000551
- Nick, F. M., Vieli, A., Howat, I. M., & Joughin, I. R. (2009). Large-scale changes in Greenland outlet glacier dynamics triggered at the terminus. *Nature Geoscience*, 2(2), 110–114. https://doi.org/10.1038/ngeo394
- Nienow, P. W., Sole, A. J., Slater, D., & Cowton, T. R. (2017). Recent Advances in Our Understanding of the Role of Meltwater in the Greenland Ice Sheet System. *Current Climate Change Reports*, 3(4), 1–15. https://doi.org/10.1007/s40641-017-0083-9
- Noël, B. P. Y., Howat, I. M., Citterio, M., & Lenaerts, J. T. M. (2017). A tipping point in refreezing accelerates mass loss of Greenland's glaciers and ice caps. *Nature Communications*, 8(1), 875. https://doi.org/10.1038/ncomms14730
- Noël, B. P. Y., van De Berg, W. J., Machguth, H., Lhermitte, S., Howat, I. M., Fettweis, X., & van den Broeke, M. R. (2016). A daily, 1km resolution data set of downscaled Greenland ice sheet surface mass balance (19582015). The Cryosphere, 10(5), 2361–2377. https:// doi.org/10.5194/tc-10-2361-2016
- Noël, B. P. Y., van De Berg, W. J., van Wessem, J. M., van Meijgaard, E., van As, D., Lenaerts, J. T. M., et al. (2018). Modelling the climate and surface mass balance of polar ice sheets using RACMO2 Part 1: Greenland (19582016). *The Cryosphere*, 12(3), 811–831. https://doi.org/10.5194/tc-12-811-2018
- Nye, J. F. (1960). The response of glaciers and ice-sheets to seasonal and climatic changes, Proceedings of the Royal Society A: Mathematical. *Physical and Engineering Sciences*, 256(1287), 559–584.
- O'Cofaigh, C., Dowdeswell, J. A., Jennings, A. E., Hogan, K. A., Kilfeather, A. A., Hiemstra, J. F., et al. (2013). An extensive and dynamic ice sheet on the West Greenland shelf during the last glacial cycle. *Geology*, 41(2), 219–222. https://doi.org/10.1130/G33759.1
- O'Leary, M., & Christoffersen, P. (2013). Calving on tidewater glaciers amplified by submarine frontal melting. *The Cryosphere*, 7(1), 119–128. https://doi.org/10.5194/tc-7-119-2013
- Oltmanns, M., Straneo, F., Moore, G. W. K., & Mernild, S. H. (2014). Strong Downslope Wind Events in Ammassalik, Southeast Greenland. Journal of Climate, 27(3), 977–993. https://doi.org/10.1175/JCLI-D-13-00067.1
- Overeem, I., Hudson, B. D., Syvitski, J. P., Mikkelsen, A. P. B., Hasholt, B., van den Broeke, M. R., et al. (2017). Substantial export of suspended sediment to the global oceans from glacial erosion in Greenland. *Nature Geoscience*, 10(11), 859–863. https://doi.org/10.1038/ ngeo3046
- Pattyn, F. (2018). The paradigm shift in Antarctic ice sheet modelling. *Nature Communications*, 9(2728). https://doi.org/10.1038/s41467 -018-05003-z
- Pattyn, F., Ritz, C., Hanna, E., Asay-Davis, X., DeConto, R., Durand, G., et al. (2018). The Greenland and Antarctic ice sheets under 1.5 °C global warming. *Nature Climate Change*, 8(12), 1053–1061. https://doi.org/10.1038/s41558-018-0305-8
- Pfeffer, W. T. (2007). A simple mechanism for irreversible tidewater glacier retreat. Journal of Geophysical Research, 112, F03S25. https://doi.org/10.1029/2006JF000590
- Phillips, T., Rajaram, H., & Steffen, K. (2010). Cryo-hydrologic warming: A potential mechanism for rapid thermal response of ice sheets. *Geophysical Research Letters*, 37, L20503. https://doi.org/10.1029/2010GL044397
- Poinar, K., Joughin, I. R., Lenaerts, J. T. M., & van den Broeke, M. R. (2016). Englacial latent-heat transfer has limited influence on seaward ice flux in western Greenland. *Journal of Glaciology*, 63(237), 1–16. https://doi.org/10.1017/jog.2016.103
- Porter, C., Morin, P., Howat, I. M., Noh, M.-J., Bates, B., Peterman, K., et al. (2018). ArcticDEM. *Harvard Dataverse*. https://doi.org/10. 7910/DVN/OHHUKH
- Post, A., O'Neel, S., Motyka, R. J., & Streveler, G. (2011). A complex relationship between calving glaciers and climate, Eos. *Transactions of the American Geophysical Union*, 92(37), 305–306.
- Price, S. F., Payne, A., Howat, I. M., & Smith, B. E. (2011). Committed sea-level rise for the next century from Greenland ice sheet dynamics during the past decade. *Proceedings of the National Academy of Sciences*, 108(22), 8978–8983. https://doi.org/10.1073/pnas.1017313108
- Pritchard, H. D., Arthern, R. J., Vaughan, D. G., & Edwards, L. A. (2009). Extensive dynamic thinning on the margins of the Greenland and Antarctic ice sheets. *Science*, 461(7), 971–975. https://doi.org/10.1038/nature08471
 Pathmann, N. M., Hvidhorg, C. S. Schward, A. M. Crinsted, A. Cudmundsson, C. H. Langen, P. L. et al. (2017). Highly temporally.
- Rathmann, N. M., Hvidberg, C. S., Solgaard, A. M., Grinsted, A., Gudmundsson, G. H., Langen, P. L., et al. (2017). Highly temporally resolved response to seasonal surface melt of the Zachariae and 79N outlet glaciers in northeast Greenland. *Geophysical Research Letters*, 44, 9805–9814. https://doi.org/10.1002/2017GL074368



- Rennermalm, A. K., Moustafa, S. E., Mioduszewski, J., Chu, V. W., Forster, R. R., Hagedorn, B., et al. (2013). Understanding Greenland ice sheet hydrology using an integrated multi-scale approach. *Environmental Research Letters*, 8(1), 015,017. https://doi.org/10.1088/ 1748-9326/8/1/015017
- Rignot, E. J., Fenty, I., Xu, Y., Cai, C., & Kemp, C. (2015). Undercutting of marine-terminating glaciers in West Greenland. Geophysical Research Letters, 42, 5909–5917. https://doi.org/10.1002/2015GL064236
- Rignot, E. J., & Kanagaratnam, P. (2006). Changes in the velocity structure of the Greenland Ice Sheet. *Science*, 311, 986–990. https://doi.org/10.1126/science.1121381
- Rignot, E. J., Koppes, M. N., & Velicogna, I. (2010). Rapid submarine melting of the calving faces of West Greenland glaciers. Nature Geoscience, 3(3), 187–191. https://doi.org/10.1038/ngeo765
- Rignot, E. J., Xu, Y., Menemenlis, D., Mouginot, J., Scheuchl, B., Li, X., et al. (2016). Modeling of ocean-induced ice melt rates of five west Greenland glaciers over the past two decades. *Geophysical Research Letters*, 43, 6374–6382. https://doi.org/10.1002/2016GL068784
- Robel, A. A. (2017). Thinning sea ice weakens buttressing force of iceberg mélange and promotes calving. *Nature Communications*, *8*, 14596. https://doi.org/10.1038/ncomms14596
- Rosenau, R., Scheinert, M., & Dietrich, R. (2015). A processing system to monitor Greenland outlet glacier velocity variations at decadal and seasonal time scales utilizing the Landsat imagery. *Remote Sensing of Environment*, 169, 1–19. https://doi.org/10.1016/j.rse.2015.07. 012
- Scambos, T. A., Bohlander, J., Shuman, C. A., & Skvarca, P. (2004). Glacier acceleration and thinning after ice shelf collapse in the Larsen B embayment, Antarctica. *Geophysical Research Letters*, 31, L18402. https://doi.org/10.1029/2004GL020670
- Scambos, T. A., Fricker, H. A., Liu, C.-C., Bohlander, J., Fastook, J., Sargent, A., et al. (2009). Ice shelf disintegration by plate bending and hydro-fracture: Satellite observations and model results of the 2008 Wilkins ice shelf break-ups. *Earth and Planetary Science Letters*, 280, 51–60. https://doi.org/10.1016/j.epsl.2008.12.027
- Sciascia, R., Straneo, F., Cenedese, C., & Heimbach, P. (2013). Seasonal variability of submarine melt rate and circulation in an East Greenland fjord. Journal of Geophysical Research: Ocean, 118, 2492–2506. https://doi.org/10.1002/jgrc.20142
- Schoof, C. G., Davis, A. D., & Popa, T. V. (2017). Boundary layer models for calving marine outlet glaciers. *The Cryosphere*, 11(5), 2283–2303. http://doi.org/10.5194/tc-11-2283-2017
- Seale, A., Christoffersen, P., Mugford, R. I., & O'Leary, M. (2011). Ocean forcing of the Greenland Ice Sheet: Calving fronts and patterns of retreat identified by automatic satellite monitoring of eastern outlet glaciers. *Journal of Geophysical Research*, 116, F03013. https://doi. org/10.1029/2010JF001847
- Seroussi, H., Morlighem, M., Rignot, E. J., Mouginot, J., Larour, E. Y., Schodlok, M. P., & Khazendar, A. (2014). Sensitivity of the dynamics of Pine Island Glacier, West Antarctica, to climate forcing for the next 50 years. *The Cryosphere*, 8(5), 1699–1710. https://doi.org/10.5194/tc-8-1699-2014
- Shapero, D. R., Joughin, I. R., Poinar, K., Morlighem, M., & Gillet-Chaulet, F. (2016). Basal resistance for three of the largest Greenland outlet glaciers. Journal of Geophysical Research: Earth Surface, 121, 168–180. https://doi.org/10.1002/(ISSN)2169-9011
- Slater, D., Goldberg, D. N., Nienow, P. W., & Cowton, T. R. (2016). Scalings for submarine melting at tidewater glaciers from buoyant plume theory. Journal of Physical Oceanography, 46, 1839–1855. https://doi.org/10.1175/JPO-D-15-0132.s1
- Slater, D., Nienow, P. W., Cowton, T. R., Goldberg, D. N., & Sole, A. J. (2015). Effect of near-terminus subglacial hydrology on tidewater glacier submarine melt rates. *Geophysical Research Letters*, 42, 2861–2868. https://doi.org/10.1002/2014GL062494
- Slater, D., Nienow, P. W., Goldberg, D. N., Cowton, T. R., & Sole, A. J. (2017). A model for tidewater glacier undercutting by submarine melting. *Geophysical Research Letters*, 380, 2360–2368. https://doi.org/10.1002/2016GL072374
- Slater, D., Straneo, F., Das, S. B., Richards, C. G., Wagner, T. J. W., & Nienow, P. W. (2018). Localized Plumes Drive Front-Wide Ocean Melting of A Greenlandic Tidewater Glacier. *Geophysical Research Letters*, 45, 12,350–12,358. https://doi.org/10.1029/2018GL080763
- Slater, D. A., Straneo, F., Felikson, D., Little, C. M., Goelzer, H., Fettweis, X., & Holte, J. (2019). Estimating Greenland tidewater glacier retreat driven by submarine melting. *The Cryosphere*, 13(9), 2489–2509. https://doi.org/10.5194/tc-13-2489-2019
- Smith, L. C., Yang, K., Pitcher, L. H., Overstreet, B. T., Chu, V. W., Rennermalm, A. K., et al. (2017). Direct measurements of meltwater runoff on the Greenland ice sheet surface. *Proceedings of the National Academy of Sciences*, 114(50), E10622–E10631. https://doi.org/10. 1073/pnas.1707743114
- Sole, A. J., Nienow, P. W., Bartholomew, I. D., Mair, D. W. F., Cowton, T. R., Tedstone, A. J., & King, M. A. (2013). Winter motion mediates dynamic response of the Greenland Ice Sheet to warmer summers. *Geophysical Research Letters*, 40, 3940–3944. https://doi.org/10.1002/ grl.50764
- Spall, M. A., Jackson, R. H., & Straneo, F. (2017). Katabatic Wind-Driven Exchange in Fjords. Journal of Geophysical Research: Oceans, 122, 8246–8262. https://doi.org/10.1002/2017JC013026
- Stearns, L. A., & van der Veen, C. J. (2018). Friction at the bed does not control fast glacier flow. *Science*, 361(6399), 273–277. https://doi. org/10.1126/science.aat2217
- Stearns, L. A., & van der Veen, C. J. (2019). Response to Comment on "Friction at the bed does not control fast glacier flow". *Science*, 363(6427), eaau8375. https://doi.org/10.1126/science.aau8375
- Stevens, L. A., Hewitt, I. J., Das, S. B., & Behn, M. D. (2018). Relationship Between Greenland Ice Sheet Surface Speed and Modeled Effective Pressure. Journal of Geophysical Research: Earth Surface, 123, 2258–2278. https://doi.org/10.1029/2017JF004581
- Straneo, F., & Cenedese, C. (2015). The Dynamics of Greenland's Glacial Fjords and Their Role in Climate. *Annual Review of Marine Science*, 7(1), 89–112. https://doi.org/10.1146/annurev-marine-010213-135133
- Straneo, F., Curry, R. G., Sutherland, D. A., Hamilton, G. S., Cenedese, C., Våge, K., & Stearns, L. A. (2011). Impact of fjord dynamics and glacial runoff on the circulation near Helheim Glacier. *Nature Geoscience*, 4(5), 322–327.
- Straneo, F., Hamilton, G. S., Stearns, L. A., & Sutherland, D. A. (2016). Connecting the Greenland Ice Sheet and the ocean: a case study of Helheim Glacier and Sermilik Fjord. *Oceanography*, 29(4), 34–45. https://doi.org/10.5670/oceanog.2016.97
- Straneo, F., Hamilton, G. S., Sutherland, D. A., Stearns, L. A., Davidson, F. J. M., Hammill, M. O., et al. (2010). Rapid circulation of warm subtropical waters in a major glacial fjord in East Greenland. *Nature Geoscience*, 3, 182–186. https://doi.org/10.1038/ngeo764
- Straneo, F., & Heimbach, P. (2013). North Atlantic warming and the retreat of Greenland's outlet glaciers. *Nature*, 504(7478), 36–43. https://doi.org/10.1038/nature12854
- Straneo, F., Heimbach, P., Sergienko, O., Hamilton, G. S., Catania, G., Griffies, S. M., et al. (2013). Challenges to Understanding the Dynamic Response of Greenland's Marine Terminating Glaciers to Oceanic and Atmospheric Forcing. Bulletin of the American Meteorological Society, 94(8), 1131–1144. https://doi.org/10.1175/BAMS-D-12-00100.1
- Straneo, F., Sutherland, D. A., Holland, D. M., Gladish, C., Hamilton, G. S., Johnson, H. L., et al. (2012). Characteristics of ocean waters reaching Greenland's glaciers. *Annals of Glaciology*, 53(60), 202–210. https://doi.org/10.3189/2012AoG60A059



- Straneo, F., Sutherland, D. A., Stearns, L. A., Catania, G. A., Catania, G., Heimbach, P., et al. (2019). The case for a sustained Greenland Ice sheet-Ocean Observing System (GrIOOS). Frontiers in Earth Science, 6(138), 391. https://doi.org/10.3389/fmars.2019.00138
- Studinger, M., Koenig, L., Martin, S., & Sonntag, J. (2010). Operation IceBridge: Using instrumented aircraft to bridge the observational gap between ICESat and ICESat-2. In *IGARSS 2010-2010 IEEE International Geoscience and Remote Sensing Symposium* (pp. 1918–1919). Honolulu HI, USA: IEEE.
- Sutherland, D. A., Jackson, R. H., Kienholz, C., Amundson, J. M., Dryer, W. P., Duncan, D., et al. (2019). Direct observations of submarine melt and subsurface geometry at a tidewater glacier. *Science*, 365(6451), 369–374. https://doi.org/10.1126/science.aax3528
- Sutherland, D. A., & Pickart, R. S. (2008). The East Greenland Coastal Current: Structure, variability, and forcing. *Progress in Oceanography*, 78(1), 58–77. https://doi.org/10.1016/j.pocean.2007.09.006
- Tedstone, A. J., Nienow, P. W., Gourmelen, N., Dehecq, A., Goldberg, D. N., & Hanna, E. (2015). Decadal slowdown of a land-terminating sector of the Greenland Ice Sheet despite warming. *Nature*, 526(7575), 692–695. https://doi.org/10.1038/nature15722
- The IMBIE team (2018). Mass balance of the Antarctic Ice Sheet from 1992 to 2017. Nature, 558(7709), 219-222. https://doi.org/ 10.1038/s41586-018-0179-y
- Todd, J., Christoffersen, P., Zwinger, T., Råback, P., Chauché, N., Benn, D., et al. (2018). A Full-Stokes 3-D Calving Model Applied to a Large Greenlandic Glacier. Journal of Geophysical Research, 308, 1–23. https://doi.org/10.1002/2017JF004349
- Truffer, M., & Echelmeyer, K. A. (2003). Of isbræ and ice streams. Annals of Glaciology, 36, 66-72. https://doi.org/10.3189/ 172756403781816347
- Truffer, M., & Motyka, R. J. (2016). Where glaciers meet water: Subaqueous melt and its relevance to glaciers in various settings. Reviews of Geophysics, 54, 220–239. https://doi.org/10.1002/(ISSN)1944-9208
- Trusel, L. D., Das, S. B., Osman, M. B., Evans, M. J., Smith, B. E., Fettweis, X., et al. (2018). Nonlinear rise in Greenland runoff in response to post-industrial Arctic warming. *Nature*, 564(7734), 1–18. https://doi.org/10.1038/s41586-018-0752-4
- Våge, K., Pickart, R. S., Sarafanov, A., Knutsen, Ø., Mercier, H., Lherminier, P., et al. (2011). The Irminger Gyre Circulation, convection, and interannual variability. Deep-Sea Research Part I, 58(5), 590–614. https://doi.org/10.1016/j.dsr.2011.03.001
- Van Wychen, W., Davis, J., Burgess, D. O., Copeland, L., Gray, L., Sharp, M. J., & Mortimer, C. (2016). Characterizing interannual variability of glacier dynamics and dynamic discharge (19992015) for the ice masses of Ellesmere and Axel Heiberg Islands Nunavut, Canada. *Journal of Geophysical Research: Earth Surface*, 121, 39–63. https://doi.org/10.1002/(ISSN)2169-9011
- Vaughan, D. G., Comiso, J. C., Allison, I., Carrasco, J., Kaser, G., Kwok, R., Mote, P., Murray, T., Paul, F., Ren, J., Rignot, E., Solomina, O., Steffen, K., & Zhang, T. (2014). Observations: Cryosphere. In: Climate Change 2013: The Physical Sci-ence Basis. *Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change* [T. F. Stocker, D. Qin, G.-K. Plattner, M. Tignor, S. Allen, A. Boschung, A. Nauels, Y. Xia, V. Bex, & P. Midgley (Eds.)], Cambridge, United Kingdom and New York, NY, USA: Cambridge Univ Press.
- van der Veen, C. J. (1998). Fracture mechanics approach to penetration of bottom crevasses on glaciers. *Cold Regions Science and Technology*, 27, 213–223.
- van der Veen, C. J. (2001). Greenland ice sheet response to external forcing. Journal of Geophysical Research, 106(D24), 34,047-34,058.

van der Veen, C. J., Plummer, J., & Stearns, L. A. (2011). Controls on the recent speed-up of Jakobshavn Isbrae, West Greenland. Journal of Glaciology, 57(204), 770–782.

van der Veen, C. J., & Whillans, I. M. (1989). Force budget: II Application to two-dimensional flow along Byrd Station Strain Network, Antarctica. *Journal of Glaciology*, 35(119), 61–67.

Velicogna, I. (2009). Increasing rates of ice mass loss from the Greenland and Antarctic ice sheets revealed by GRACE. Geophysical Research Letters, 36, L19503. https://doi.org/10.1029/2009GL040222

Velicogna, I., Sutterley, T. C., & van den Broeke, M. R. (2014). Regional acceleration in ice mass loss from Greenland and Antarctica using GRACE time-variable gravity data. Geophysical Research Letters, 41, 8130–8137. https://doi.org/10.1002/2014GL061052

- Vernon, C. L., Bamber, J. L., Box, J. E., van den Broeke, M. R., Fettweis, X., Hanna, E., & Huybrechts, P. (2013). Surface mass balance model intercomparison for the Greenland ice sheet. *The Cryosphere*, 7(2), 599–614. https://doi.org/10.5194/tc-7-599-2013
- Wagner, T. J. W., Straneo, F., Richards, C. G., Slater, D. A., Stevens, L. A., Das, S. B., & Singh, H. (2019). Large spatial variations in the flux balance along the front of a Greenland tidewater glacier. *The Cryosphere*, 13(3), 911–925. https://doi.org/10.5194/tc-13-911-2019
- Walter, F., Chaput, J., & Lüthi, M. P. (2014). Thick sediments beneath Greenland's ablation zone and their potential role in future ice sheet dynamics. *Geology*, 42(6), 487–490. https://doi.org/10.1130/G35492.1
- Warren, C. R., & Glasser, N. F. (1992). Contrasting Response of South Greenland Glaciers to Recent Climatic Change. Arctic and Alpine Research, 24(2), 124–132.
- Weertman, J. (1961). Stability of ice-age ice sheets. Journal of Geophysical Research, 66(11), 3783-3792. https://doi.org/10.1029/JZ066i011p03783
- Werder, M. A., Hewitt, I. J., Schoof, C. G., & Flowers, G. E. (2013). Modeling channelized and distributed subglacial drainage in two dimensions. Journal of Geophysical Research: Earth Surface, 118, 2140–2158. https://doi.org/10.1002/jgrf.20146
- Wilkinson, M. D., Dumontier, M., Aalbersberg, I. J., Appleton, G., Axton, M., Baak, A., et al. (2016). Comment: The FAIR Guiding Principles for scientific data management and stewardship. *Nature Publishing Group*, *3*(1), 1–9. https://doi.org/10.1038/sdata.2016.18
- Wilson, N., Straneo, F., & Heimbach, P. (2017). Satellite-derived submarine melt rates and mass balance (20112015) for Greenland's largest remaining ice tongues. *The Cryosphere*, *11*(6), 2773–2782. https://doi.org/10.5194/tc-11-2773-2017
- Xu, Y., Rignot, E. J., Menemenlis, D., & Koppes, M. N. (2012). Numerical experiments on subaqueous melting of Greenland tidewater glaciers in response to ocean warming and enhanced subglacial discharge. *Annals of Glaciology*, 53(60), 229–234. https://doi.org/10. 3189/2012AoG60A139
- Young, T. J., Christoffersen, P., Doyle, S. H., Nicholls, K. W., Stewart, C. L., Hubbard, B., et al. (2019). Physical Conditions of Fast Glacier Flow: 3 Seasonally-Evolving Ice Deformation on Store Glacier, West Greenland. *Journal of Geophysical Research: Earth Surface*, 124, 245–267. https://doi.org/10.1029/2018JF004821
- Zhan, Z. (2019). Seismic Noise Interferometry Reveals Transverse Drainage Configuration Beneath the Surging Bering Glacier. *Geophysical Research Letters*, 46, 4747–4756. https://doi.org/10.1029/2019GL082411
- Zwally, H. J., Abdalati, W., Herring, T., Larson, K. M., Saba, J. L., & Steffen, K. (2002). Surface Melt-Induced Acceleration of Greenland Ice-Sheet Flow. Science, 297(5579), 218–222.